ASSESSMENT OF SELECTED TYPES OF SOIL DEGRADATION IN SYFERKUIL FARM USING REMOTE SENSING TECHNIQUES

ΒY

MAMOROBELA, KARABO EDWIN

A MINI-DISSERTATION

SUBMITTED IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE

OF

MASTER OF SCIENCE

IN

SOIL SCIENCE

IN THE

FACULTY OF SCIENCE AND AGRICULTURE

(SCHOOL OF AGRICULTURAL AND ENVIRONMENTAL SCIENCES)

AT THE

UNIVERSITY OF LIMPOPO

Supervised by

Dr. M.G. ZERIZGHY

MARCH, 2023

TABLE OF CONTENTS	ΤΙΤΙ Ε ΡΔΩΕ
	iii
DEDICATION	
ACKNOWLEDGEMENTS	iv
ABREVIATIONS AND ACRONYMS	v
LIST OF TABLES	vi
LIST OF FIGURES	vii
ABSTRACT	viii
CHAPTER 1: THE GENERAL BACKGROUND	1
1.1 Introduction	1
1.2 Purpose of the study	2
1.2.1 Aim	2
1.2.1 Objective(s)	2
1.3 Motivation of the study	2
1.4 The general structure of the thesis	
CHAPTER 2: ASSESSING SOIL SURFACE SALT ACCUMULATION USIN	G
MULTISPECTRAL IMAGES IN SYFERKUIL FARM	5
Abstract	5
2.1 Background	7
2.2 Problem statement	
2.3 Literature review	
2.3.1 Work done on problem statement	
2.3.2 Work not done on problem statement	
2.4 Research methodology	
2.4.1 Description of the study area	
2.4.2 Data collection	
2.4.2.1 Conventional methods	

2.4.2.2 Remote sensing	
2.4.3 Data analysis	17
2.5 Results	18
2.5.1 Conventional laboratory analysis results for soil salinity	
2.5.2 Prediction of soil salinity using PLSR model	19
2.6 Discussion	
2.6.1 Conventional laboratory analysis results for soil salinity	
2.6.2 The performance of model developed from multispectral satellite image	21
2.7 Conclusion and recommendations	23
CHAPTER 3: ASSESSMENT OF RILL DEGRADED SOILS IN SYFERKUIL FARM USING REMOTE SENSING TECHNIQUES	
Abstract	24
3.1 Background	
3.2 Problem statement	26
3.3 Literature review	
3.3.1 Work done on problem statement	
3.3.2 Work not done on problem statement	
3.5 Research methodology	
3.5.1 Description of the study area	
3.5.2 Data collection	31
3.5.2.1 Remote sensing	31
3.5.3 Data analysis	
3.6 Results	34
3.6.1 Supervised image classification results	
3.6.2 Accuracy results	36
3.7 Discussion	
3.8 Conclusion and recommendations	40

CHAPTER 4: ASSESSING SOIL ORGANIC CARBON DEPLECTION USING	
MULTISPECTRAL IMAGES IN SYFERKUIL FARM	41
Abstract	41
4.1 Background	42
4.2 Problem statement	
4.3 Literature review	44
4.3.1 Work done on problem statement	
4.3.2 Work not done on problem statement	
4.4 Research methodology	47
4.4.1 Description of the study area	47
4.4.2 Data collection	
4.4.2.1 Conventional methods	
4.4.2.2 Remote sensing	
4.4.3 Data analysis	
4.5 Results	
4.5.1 Conventional laboratory analysis results for soil organic carbon (SOC) and	
soil organic matter (SOM)	
4.5.2 Prediction of soil organic carbon using PLSR model	
4.6 Discussion	
4.6.1 Soil organic carbon and soil degradation forms	
4.6.2 The performance of model developed from multispectral satellite image	
4.7 Conclusion and recommendations	
CHAPTER 5: SUMMARY AND RECOMMENDATIONS	
5.1 Summary	
5.2 Recommendations	
REFERENCES	
LIST OF APPENDICES	70

DECLARATION

I, Mamorobela Karabo, declare that the mini-dissertation hereby submitted to the University of Limpopo, for the degree of Bachelor of Science in Agriculture (Soil science) has not previously been submitted by me for a degree at this or any other University; that it is my work in design and in execution, and all material contained herein has been duly acknowledged.

23 March 2023

MAMOROBELA K.E (Mr)

Date

DEDICATION

This study is dedicated to my lovely parents Mr. and Mrs. Mamorobela and my three siblings, Malego, Matome and Stanley Mamorobela.

ACKNOWLEDGEMENT

I would like to acknowledge my supervisor Dr. M.G Zerizghy who has been helping me with this dissertation from the beginning. I would not have managed to successfully complete this research without his assistance. He led me in the right direction where I seemed to go astray, and he never gave up on me. I would like to thank my parents Mr. and Mrs. Mamorobela and my three siblings for the unconditional support, love and their endless prayers and for motivating me to show dedication in everything I do. I would also like to acknowledge the University of Limpopo, School of Agricultural and Environmental Sciences for providing us resources whenever we visited the site where this research was conducted. I'm grateful for the funding that National Research Foundation (NRF) provided for me throughout my research. Finally, I would like to thank The Lord Almighty for being my rock, pillar and foundation. All praise and thanks unto God.

ABBREVIATIONS AND ACRONYMS

CEC	Cation Exchange Capacity
dS/m	DeciSiemens per metre
EC	Electrical conductivity
ESA	European Space Agency
FAO	Food and Agricultural Organization
GPS	Global positioning system
Landsat	Land Remote Sensing Satellite (system)
mS /cm	MilliSiemens per centimetre
PLSR	Partial Least Square Regression
QGIS	Quantum Geographic Information System
R ²	Coefficient of determination
RF	Random Forest
RIO	Region of Interest
RMSE	Root Mean Square Error
RPD	Relative Percentage Deviation
RS	Remote Sensing
SCP	Semi-Automatic Classification Plugin
SD	Standard Deviation
SOC	Soil Organic Carbon
SOM	Soil Organic Matter
SWIR	Short-Wave Infrared
VNIR	Visible Near-Infrared

LIST OF TABLES

Table 2.1 : Classification of soil based on the electrical conductivity (EC) classification	15
Table 2.2 : Prediction of goodness of the model based on the coefficient of determination	18
Table 2.3 : Prediction of goodness of the model based on the relative percentage	18
Table 2.4 : Statistical description of soil EC analysed by conventional laboratory methods (n= 55) in Syferkuil farm, Limpopo	19
Table 2.5 : PLSR model performance for predicting soil EC using multispectral satellite in Syferkuil farm, Limpopo, South Africa	20
Table 3.1 : Showing Showing the classification to interpret the strength of agreement based on the Cohen's Kappa value	t 36
Table 3.2 : Showing Error Matrix Code, Reference, Classified and Pixel Sum values	
Table 3.3 : Worldview 2 satellite image accuracies(%) for rill and no erosion	38
Table 4.1 : Classification of soil based on the SOC	48
Table 4.2 : Classification of soil based on the SOM content	
Table 4.3 : Prediction of goodness of the model based on the coefficient of determination (R2)	51
Table 4.4 : Prediction of goodness of the model based on the relative percentage deviation (RPD)	
Table 4.5 : Statistical description of soil organic carbon analysed by conventional laboratory methods (n =98) in Syferkuil farm, Limpopo, South Africa	53
Table 4.6 : PLSR model performance for soil organic carbon prediction	54

LIST OF FIGURES

Figure 1.1 : The general structure of the thesis	4
Figure 2.1 : Study Area and sampling points for extraction of spectral measuremen	nts14
Figure 2.2 : Flow chart indicating methodology for estimating soil salinity	
Figure 3.1 : Study site in the syferkuil farm, Limpopo	
Figure 3.2 : Flow charts showing supervised image classification	
Figure 3.3 : The cultivated area with the selected training areas	
Figure 3.4 : Randomly selected validation points	
Figure 3.5 : Classification done with spectral angle algorithm	
Figure 4.1 : Study Area and sampling points for soil organic carbon	
Figure 4.2 : Flowchart of methodology used in this study	

ABSTRACT

Soil degradation is a serious environmental threat facing the humanity today. Soil degradation in one or more of its forms has been labelled as a 'global pandemic'. This is because; soil degradation is a very serious world problem and affects all countries or continents. Thus there is an acute need to devise a way of reducing its vast advance. This is why, it is important to establish the magnitude and the extent of soil degradations in order to mitigate its effect. The objectives of this study were to: (i) Identify soils degraded by rill erosion with acceptable accuracy from remote sensing images. (ii) Determine soil organic carbon status with acceptable accuracy from remote sensing. (iii) Determine soil surface salt accumulation with acceptable accuracy from remote sensing images. The soil degradation forms considered in this study are soil salinity, rill soil erosion and soil organic carbon. Nutrient depletion is another significant chemical process of soil degradation. Soil organic carbon depletion is a chemical degradation and in most instances is influenced by human and natural activities. The assessment of these soil degradation forms has been done in three separate chapters and detailed abstract is given at the start of each chapter. However, the general findings revealed that the prediction of those soil degradation forms from remotely sensed images did not yield good results. Nonetheless, promising performance has been recorded. Recommendations for feature studies are also provided.

Keywords : Soil degradations, Remote sensing, Multispectral imagery (sentinel 2), Regression analysis, Syferkuil farm.

CHAPTER ONE

1. THE GENERAL BACKGROUND

1.1 INTRODUCTION

Soil degradation is the most serious global environmental issue, which is why it is considered as a global pandemic, it is world problem. Soil degradation processes involves displacement of soil materials and internal soil physical and chemical deteriotation. Soil degradation is influenced by natural and human activities. However, soil degradation processes vary according to the land uses and management (DeLong *et al.*, 2015; Gopalakrishnan and Kumar, 2020). Soil degradation, mainly occurs in three forms namely, physical, chemical, and biological soil degradation. The physical degradation include (i.e. soil erosion, specifically rill erosion) and chemical degradations (i.e. depletion of organic carbon and salinization). In South Africa, soil degradations are predominant. Soil degradations pose a threat on agricultural land particularly in arid and semi-arid regions (Ren *et al.*, 2019). The dominating factors that initiate the development of soil degradation are accelerated by the soil type, topography, and climate (Lal, 2015).

In agriculture, soil degradation poses a threat to food security, food production and environment conservation. This can then force farmers to look for new productive land. Furthermore, monitoring soil quality degradation is a crucial step in practicing precision agriculture and making informed decisions with regards to the future land use and management for reclamation and rehabilitation of soils (Peng *et al.*, 2019). Therefore, it is of crucial importance to realise that soils are our most essential resources meaning that they must be used, improved and restored. In this study, remote sensing (RS) which is defined as the process of gathering information about an object through the use of electromagnetic radiation, from distance, without making physical contact with the object itself becomes a crucial tool for assessing, monitoring, and determining soil degradation forms (Chauhan, 2015 and Paterson *et al.*, 2015). Thus, this study aims to establish if soil degradation forms could be identified and quantified using remote sensing as an alternative method to the conventional methods (i.e., field work or survey). Remote sensing techniques are able to provide rapid analysis of soil information and covering whole land surface at an acceptable level of details (Paterson *et al.*, 2015). In addition, remote sensing provides simultaneous collection of data systematically and non-requirement of chemical reagents (Chauhan, 2015). Remote sensing has the potential to reveal information about soils since the signals measured are in relation to the physical measures, which can be linked to soil properties (Mirzabaev *et al.*, 2016; Rossel *et al.*, 2011; and Zhou *et al.*, 2015).

1.2 PURPOSE OF THE STUDY

1.2.1 Aim

The realistic assessment of selected soil degradation forms in the Syferkuil farm.

1.2.2 Objectives

The objectives of this study are to:

- i) Identify soils degraded by rill erosion with acceptable accuracy from remote sensing images.
- ii) To determine soil organic carbon status with acceptable accuracy from remote sensing.
- iii) To determine soil surface salt accumulation with acceptable accuracy from remote sensing images.

1.3 MOTIVATION OF THE STUDY

This study will be crucial in terms of increasing awareness on the importance of managing and restoring soil quality to minimise the risk of soil degradation forms. The results from this study will be crucial in land use management studies and soil reclamation programmes for environmental protection. Moreover, this study will help farmers to gain good understanding of the status of the soil which will help in coming up with precise management strategies for reclamation and rehabilitation of soils. The premise of maintaining and rehabilitating soil quality starts by establishing or assessing the existing condition (Lal, 2015). Hillel *et al.* (2015) pointed out that we cannot protect what we don't understand. Therefore, the results of this study will provide a better understanding of assessing soil degradations using remote sensing.

Remote sensing techniques as an alternative tool to the conventional methods are able to assess and monitor selected types of soil degradations with acceptable accuracy (Paterson *et al.*, 2015). This is due to rapid analysis of soil information and covering whole land surface at an acceptable level of details for monitoring spatial variability of soil degradation forms. Conventional methods tend to be ineffective when dealing with the processing of large soil information due to their long procedure that delays the analysis of soil parameters (Chauhan, 2015).

According to Paterson *et al.* (2015), there is a shortage of up to date soil information about South African soils. Furthermore, there is little work done that utilizes remote sensing for assessing soil degradation in South Africa particularly in the Limpopo Province. However, majority of studies are undertaken in KwaZulu Natal and Eastern Cape Province (Mararakanye, 2015). Factors that influence the development of soil degradation tend to differ in regions and areas (Sonneveld *et al.*, 2005; Nazari Samani *et al.*, 2009). Additionally, the performance of the Partial Least Square Regression (PLSR) model varies with the location or study area and the soil parameter being investigated or assessed. Moreover, the need to study the use of remote sensing is important because the available literature mostly focused on the main factors that accelerate soil degradation rather than developing models that will assist in assessing soil degradation. This study will focus on accurate assessment of soil degradation using remote sensing techniques.

1.4 THE GENERAL STRUCTURE OF THE THESIS

The upcoming three chapters address one objective each and a final chapter that provides general summary at the end of the thesis (Figure 1.1). The content of each chapter is described briefly as follows:

Chapter one deals with the general background of soil degradation. I outlined the aim and objectives of the study. Furthermore, I explained the reasoning why this problem needs to be addressed and what will be learnt from it. Chapter two, deals with the assessment of soil surface salt accumulation (soil salinity). Chapter three, deals with the assessment of soils degraded by rill erosion. Chapter four, deals with the assessment of soil organic carbon depletion. Chapter five, deals with the summary and possible recommendations.



Figure 1.1: The general structure of the thesis.

CHAPTER TWO

ASSESSING SOIL SURFACE SALT ACCUMULATION USING MULTISPECTRAL IMAGES IN SYFERKUIL

ABSTRACT

Monitoring soil quality is an essential practice for agriculture and environmental protection. Soil salinization is one of the most globally significant environmental soil hazards, which results in severe land degradation and desertification. This is actually caused by both natural and human activities. The excess accumulation of soluble salts in the soil surface which basically referred to as salinization has severe impact on agricultural production, biodiversity and sustainable development. The objective of this study was to determine soil surface salt accumulation with acceptable accuracy from remote sensing images. The study was conducted in Syferkuil, experimental farm of the University of Limpopo. Fifty five soil samples were collected and a grid sampling of 50 m by 50 m was followed. Global positioning system (GPS) was used to record the exact location of each sampled point in each grid. The electrical conductivity (EC) of the soil samples was determined conventionally in the laboratory using Mettler Toledo EC meter. The standard laboratory (conventional) methods for determining soil texture and, pH were used. A multispectral image (Sentinel 2) with more than 10 bands ranging from the visible to shortwave infrared was used to predict the soil EC. The coordinates of the sampling points were used to extract spectral value from the image. The spectral values and the conventionally determined soil EC were modelled using partial least square regression (PLSR).

The PLSR model yielded a coefficient of determination (R^2) = 0.468 and the Root Mean Square Error (RMSE) = 0.44 ds/m. The result shows that approximately half of the EC variation could be explained by the reflectance values as recorded in the image. The deviation of the predicted values from the PLSR analysis was minimal as indicated by the low RMSE value. The low R^2 value indicates that there are confounding factors. The confounding factors might come in the form of noise and errors due to variations in soil surface roughness, geometric and atmospheric effects. Furthermore, the performance of models might be because of low spectral resolution of the image as compared to hyperspectral data, which have bands with narrow wavelengths. Despite low prediction accuracy in this study, remote sensing techniques are efficient tools to monitor and detect soil salinity unlike in the old days, when conventionally laboratory methods were the only means of assessing soil salinity problems. Sentinel 2 could be used to make preliminary study of EC before detailed in situ assessment could be done. This is because; Sentinel 2 is able to provide excellent spatial coverage of a large area, and making it easy to obtain soil information (Gorji *et al.*, 2017). Thus, it is recommended that satellite image with better spectral resolution (hyperspectral) be investigated to see if there be an improvement in the model performance.

Keywords : Soil degradation, Soil salinity, Multispectral imagery (sentinel 2), Regression analysis,

2.1 BACKGROUND

Soil salinity is defined as the build-up of salts mainly chloride, sulfates and carbonates of sodium, calcium, and magnesium in the soil surface (Smith and Doran, 1996). It is determined by measuring soil electrical conductivity (EC) due to its high correlation with soil salinity. Monitoring soil salinity or soil quality degradation in general, is an essential practice for precise management strategies for reclamation and rehabilitation of soils. It is determined by measuring soil electrical conductivity (EC) due to its high correlation with soil salinity. Monitoring soil salinity or soil quality degradation in general, is an essential practice for precise management strategies for reclamation and rehabilitation of soils. It is determined by measuring soil salinity or soil quality degradation in general, is an essential practice for precise management strategies for soil quality degradation in general, is an essential practice for precise management strategies for soil quality degradation in general, is an essential practice for precise management strategies for reclamation and rehabilitation of soils.

In this study, one form of soil quality degradation - soil salinization is the centre of focus. It is considered as one of the leading causes of land degradation and has been found to be directly linked to the different natural processes and human activities (Gopalakrishnan and Kumar, 2020). Additionally, it has become one of the most serious global environmental issues, and as a result influencing agricultural production and food security since it leads to reduced soil and water quality especially in arid and semi-arid regions (Ren *et al.*, 2019). It predominates in these regions with the fact that evaporation rates are extremely high and water evaporates rapidly as a result leaving dissolved mineral salts in the topsoil (Pennock et al., 2015).

Despite the abundant evidence, regarding the damages that arise from salt affected soils, the problem is ever increasing rather than decreasing all over the world (Metternicht and Zinck, 2003). It should, however, be noted that assessing, monitoring and identifying areas that are mostly affected by soil degradation forms with acceptable accuracy from remote sensing is indeed a crucial step. Over the past decades, the conventional methods have been used by many soil scientists to characterise spatial and temporal variabilities of soil properties. The main difficulties of such methods lie on meeting high demands of detailed soil information in short period of time with reasonable cost (Steinberg *et al.*, 2016). Furthermore, conventional methods are time consuming and labour intensive particularly for regional level whereby data is required for large scale applications.

The complexity and cost of conventional methods in monitoring soil degradation forms has motivated the use of remote sensing techniques as an alternative method to the conventional methods. Remote sensing (RS) is defined as the process of gathering information about an object through the use of electromagnetic radiation, from distance, without making physical contact with the object itself (Chauhan, 2015). RS images can provide valuable information that might not be able to obtain using other methods since our visual perception is limited to some portions of electromagnet spectrum (EMS).

Remote sensing has the capability to cover different areas which might be difficult to cover using some other terrestrial means. In addition, remote sensing can help to observe change that might be occurring over time and to understand the spatial extent and rate of this problem. Lastly, remote sensing can provides records of soil degradation forms at those specific times the images were captured. The objective of this study was to determine soil surface salt accumulation with acceptable accuracy from remote sensing images. It is hypothesised that remote sensing images will enable accurate determination of soil surface salt accumulation.

2.2 PROBLEM STATEMENT

Soil salinity is considered as an environmental degradation and one of the leading causes of land degradation (Aldabaa *et al.*, 2015). It is a serious problem around the world due to the devastating impacts on agricultural farmlands. Thus, it is considered as a global pandemic since it affects the whole world and the problem is ever increasing rather than decreasing. Worldwide, productive and fertile soils are scarce resources and many farmers are looking at short term benefits and ignoring the long term consequences that arise from salt affected soils. In previous studies around the world, many soil scientists use conventional methods (i.e., laboratory methods or field surveys) for assessments of soil degradations. Conventional methods have drawbacks as they are time consuming and expensive. The main difficulties of conventional methods lie on meeting high demands of detailed soil information in short period of time with reasonable cost (Steinberg et al., 2016). Furthermore, it is time consuming and labour intensive particularly for regional level whereby data is required for large scale applications. For instance, with the current demand for up to date soil information particularly for regional level whereby data is required for large scale applications, these methods delay the process of acquiring necessary soil information of high accuracy in a short period of time, due to their

long procedure (Rossel et al., 2011).

Conventional methods tend to be ineffective when dealing with the large soil information due to their long procedure that delays the analysis of soil parameters. This long procedure method, however, could be addressed by the use of remote sensing techniques. This is because soil scientists have identified remote sensing as an alternative method for assessment and monitoring of soil degradation forms due to their advantages over conventional methods. These includes, simultaneous collection of data systematically and non-requirement of chemical reagents (Chauhan, 2015). Moreover, it can provide rapid analysis of soil information and covering large land surface at an acceptable level of details.

Remote sensing images can provide valuable information that might not be able to obtain using other methods since human visual perception is limited to the visible range of the of EMS (Chauhan, 2015). This means that it can provide information beyond our human visual perception. Remote sensing images can provide the records of soil degradation forms at those specific times the images were captured (Forkuor *et al.*, 2017). This can help to observe change that might be occurring over time and to understand the spatial extent and rate of this problem. Moreover, it can cover different areas which might be difficult to cover using some other terrestrial means (Chauhan, 2015).

Advantages of using remote sensing technology include saving time, wide coverage (satellite remote sensing provides the only source when data is required over large areas or regions), are faster than ground methods, and facilitate long term monitoring. These techniques provide multispectral image with resolutions that can range from medium to high, as well as hyperspectral image. These remotely sensed data have been successfully used for monitoring and mapping soil salinity for decades with mixed results (Aldabaa *et al.*, 2015; Asfaw *et al.*, 2016). Many researchers have used different techniques to monitor and map soil salinity using remote sensing data, as discussed below. A multispectral imagery is able to provide excellent spatial coverage of a large area, and making it easy to obtain soil information (Gorji *et al.*, 2017).

2.3 LITERATURE REVIEW

2.3.1 Work done on the problem statement

2.3.1.1 The importance of assessing soil surface salt accumulation.

The accumulation of salts in the soil surface tend to affect the interaction between plants and soils which in turn influence the nutrient and water availability and thus affecting crop growth and productivity (Asfaw et al., 2016; Gorji et al., 2017). This can then force famers to abandon their farmlands due to the incidence of high accumulation of salts in the soil surface. Therefore, assessing, monitoring, and mapping salt affected areas will enable better understanding of the threat posed by soil salinization in different locations. Aldabaa et al. (2015) found that soil salinity leads to reduced crop productivity and in this case, it was inherent from parent materials (i.e. where the soil is formed from). These findings were supported by Clay et al. (2001) who found that salt surface accumulation tend to have devastating impacts on plant growth and production. Despite the awareness regarding the damages that arise from surface salt accumulation on agricultural soils, the problem is ever increasing rather than declining (Metternicht and Zinck, 2003). For example, Gao et al. (2021) monitored temporal and spatial dynamics of soil salinization changes using remote sensing and Geographic information system (GIS) in China. They found an increase in salt affected areas. Fey and Mashimbye et al. (2012) also found proof of increasing soil salinity in Western Cape Province.

2.3.1.2 The use of remote sensing in assessing, monitoring, and mapping soil salinity.

Al-Gaadi *et al.* (2021) mapped soil salinity in agricultural fields in Saudi Arabia using Sentinel 2 images. They found that the relationship between EC and Sentinel 2 data showed moderate to highly significant correlations ($R^2 = 0.43 - 0.83$). In a study by Gorji *et al.* (2019), the remote sensing techniques and methods were used to assess and map soil salinity. The results of which showed that RS data provides high precision salinity maps to monitor salt affected areas. Qu *et al.* (2008) used the Partial Least Square Regression (PLSR) method to assess the salinity using hyperspectral data. The results showed that the calibrated PLSR method could predict soil salinity with precise or accurate results. In a study by Zarei *et al.* (2021), they investigated soil salinity monitoring and EC mapping using sentinel 2 satellite images. The results of this study demonstrated that the remote sensing data could provide high-precision salinity maps to monitor soil salinity as an environmental problem.

A study by Leon *et al.* (2012) aimed at predicting soil properties using Partial least square regression (PLSR) and Visible Near-Infrared (NIR) spectroscopy in the Mediterranean soils from Southern Italy have been done. The results of this study showed that PLSR is very good in predicting soil properties. In addition, the results of the PLSR were in good agreement with the correlations between soil properties and reflectance at various wavelengths. These findings are in line with those of Wenjun *et al.* (2014) which reported that partial least square regression (PLSR) is a fast analytical tool to predict soil parameters such as soil salinity, total carbon, soil moisture, and cation exchange capacity (CEC) with high accuracy. Therefore, Stenberg *et al.* (2010) suggested that remote sensing techniques are effective tools in terms of providing rapid assessment, having up to date soil information and covering large area of land at an acceptable level of detail.

A study by Goossens *et al.* (1993) aimed at examining and comparing the accuracy of multispectral sensors (Landsat TM, MSS, and SPOT) for soil salinity mapping has been done in India. They found that Landsat TM was an effective tool for soil salinity mapping. Ahmed and Andrianasolo (1997) compared the performance of the Landsat TM and SPOT XS in Pakistan. Their results were opposite to that of Goossens *et al.* (1993). They found that the SPOT XS data were more helpful than Landsat TM as it provided finer details of various thematic variables. In a study by Huang *et al.* (2005) aimed at identifying saline areas dominated by sodium chlorides and sodium sulfates using ASTER imagery. The results of this study therefore showed a good correlation between surface salt concentrations and band 1 of the ASTER sensor, followed by bands 2 and 3.

A study by Mohammad *et al.* (2019) in the Kuh Sefid village, Qom Province, Iran has been done in assessing soil salinity using Sentinel-2 multispectral imagery. They have found that the Green, Red Edge 1, SI2, SWIR2, and BI, had the best performance in model development. Moreover, they emphasized that these features could be used as optimal salinity indicators for monitoring soil salinity through satellite imagery in future studies. 2.3.1.3 Factors affecting the prediction accuracy of multispectral satellite data.

Soil moisture content tends to influence the prediction accuracy of different sensors. This is because as soil moisture increases, the reflectance of soil decreases. This actively demonstrates that soil moisture is inversely proportional to the reflectance. For instance, if there is high amount of water in the soil, the reflectance will go down. However, drier soil such as sandy textured soil reflects more than wet soils (i.e. clay textured soils after same exposure of wetness) (Kumar and Sharma, 2020). The surface roughness is also considered as one of the main factors that influence the prediction accuracy. For instance, the smaller the local surface roughness, the greater the spectral reflectance. Therefore, Gorji *et al.* (2017) and Shahabi *et al.* (2017) pointed out that the performance of the models varies with the study site and the regression techniques used. Lastly, Allbed and Kumar (2013) stated that multispectral data has limitations because of the coarse spatial and spectral resolutions.

2.3.1.4 Human and environmental parameters related to soil salinity.

Aldabaa *et al.* (2015) reported that parent materials, soil types, and topography influence soil surface salt accumulation and constrain the growth of many crops. These natural factors tend to vary with location and farm management. The application of irrigation systems that contains high amount of magnesium and calcium results in high accumulation of salts in the soil surface. This can then compromise plant productivity and soil fertility since it affects soil properties. Furthermore, application of high amount of fertilizers can lead to high accumulation of salts in the soil surface and results in land abandonment and high economic costs for soil reclamation and rehabilitation (Aldabaa *et al.*, 2015).

2.3.1.5 How soil properties are related with spectral reflectance obtained from remote sensing images.

Shrestha (2006) reported that humus, gypsum and water soluble salts are negatively correlated with spectral reflectance while carbonates are positively correlated with spectral reflectance. Visible, short wave, and near infrared bands have the ability to estimate chemical, physical, and biological soil properties (Stenberg *et al.*, 2010). Moreover, spectral analysis illustrated the high potential of short-wave infrared (SWIR) bands to identify saline soils. The results from the study conducted by

Shrestha (2006) shows very weak correlation between soil salinity and V/NIR spectra.

2.3.2 Work not done on the problem statement

Over the past decades, majority of the studies have used multispectral sensors such as Landsat, SPOT, and Advanced Space borne Thermal Emission and Reflection Radiometer (ASTER) and IKONOS in assessing soil degradation forms (Dwivedi *et al.*, 2008). These broad-band sensors provide high spatial and spectral resolution imagery. Although various studies have been conducted under different regions to assess the ability of multispectral sensor (i.e. sentinel 2) not much work has been done on the Semi-arid area of Limpopo, particularly the Mankweng area. Most of the studies have focused on the main factors that influence soil degradation rather than developing models that will help in land management studies and soil reclamation programmes since soil degradation changes with space and time. This study will establish if soil salinity could be identified and quantified using remote sensing as an alternative method to the conventional or traditional methods (i.e., laboratory methods).

2.4. RESEARCH METHODOLOGY

2.4.1 Description of the study area

The study was conducted in Syferkuil, experimental farm of the University of Limpopo (Figure 2.1). The farm is about 1,650 ha in size in general. The study was conducted on the fields reserved for student experiments. The portion of the farm that was used for this study has an area of 61 ha. For the past years the study site has been cultivated coupled with occasional resting. The soils from Syferkuil farm are identified as sandy loam texture and the dominant soil forms on the site are Shortlands and Clovelly (Soil Classification Working Group, 1991). The research site is located in semi-arid region with average rainfall of 450-630 mm per annum.



Figure 2.1: Study area and sampling points for extraction of spectral measurements

2.4.2 Data collection

2.4.2.1 Conventional methods

a) Field sampling

A grid sampling of 50 m by 50 m was followed to collect 55 soil samples. A shovel was used to take soil samples from the top 5 cm of soil surface. The soil samples are taken only in the top 5 cm soil layer for which the reflectance is thought to represent. Then samples were placed in plastic bags and were taken to soil laboratory. Global positioning system (GPS) was used to record the exact location of

each sampled point in each grid.

b) Soil physiochemical properties

Soil samples were air-dried before analyzing for soil physiochemical properties. The electrical conductivity (EC) of the soil samples was determined conventionally in the laboratory using Mettler Toledo EC meter following a method from Jones (2001). Based on the classes determined by (Durand, 1983), five salinity classes were considered as shown in Table 2.1.

Table 2.1: Classification of soil based on the electrical conductivity (EC) classification supplied by Duran (1983).

EC (ds/m)	Salinity classes
EC < 0.6	Non-saline soil
0.6 < EC >1.0	Slightly saline soil
1.0 < EC >2.0	Moderately saline
2.0 < EC >4.0	Very saline soil
EC > 4.0	Extremely saline

Note. EC = Soil electrical conductivity, (ds/m) = DeciSiemens per metre

Particle size distribution was determined using the hydrometer method (Bouyoucos, 1962). Soil pH was first measured in deionized water (1:2 soil, water) followed by 0.01 M calcium chloride (CaCl₂) using a calibrated glass electrode pH meter (Rhodes, 1982).

2.4.2.2 Remote sensing

a) Image acquisition

A multispectral image called sentinel 2 was downloaded from the European Space Agency (ESA). The image had more than 10 bands ranging from the visible to shortwave infrared and with 10 m spatial resolution. A multispectral satellite image of the study area was collected on 09 March 2021.

b) Image processing

Before the multispectral satellite image was used, it went through pre-processing in order to remove the effects of the atmosphere where radiance values received at the sensors where converted into reflectance or spectral values using Quantum Geographic Information System (QGIS). Before the multispectral satellite image was used, it went through pre-processing. This was done in order to correct any distortion inherent in the images due to the characteristics of the imaging system and conditions. The imagery was atmospherically corrected using QGIS. The coordinates of the sampling points were used in QGIS to extract spectral or reflectance values from the image using vector point extraction. The methodology for estimating soil salinity includes; the collection of soil samples from the study area and laboratory analysis to determine the soil physiochemical properties; determining the reflectance values of sentinel 2 bands ;and determining the relationship between the spectral values extracted from the Sentinel 2 imagery and the conventionally determined soil physiochemical properties(Figure 2.2).



Figure 2.2: Flowchart indicating methodology for estimating the soil salinity in the Syferkuil farm, Limpopo, South Africa.

2.4.3 Data analysis

The soil samples were divided into calibration and validation group. Therefore, 70% was used for training or model development and 30% for testing or model validation. This study used one regression model which is Partial Least Square Regression (PLSR) model to estimate the relationship or correlation between the corresponding pixels or spectral values extracted from multispectral satellite imagery (Sentinel 2) and conventionally determined soil EC. The software used for PLSR is XLSTAT 2014. PLSR was chosen due to its capability to analyse large data, and it is more interpretable (Asfaw *et al.*, 2016). It is an effective tool for assessment of soil surface salt accumulation (Gorji *et al.*, 2016).

2.4.3.1 Calibration

Conventional laboratory measurement and their corresponding reflectance were used for calibrating a partial least square regression (PLSR) model. More than two thirds of the data were used in this exercise to select the spectral bands that provide best prediction of soil EC.

2.4.3.2 Validation

Validation of the developed models was done with independent set of data that was not used for calibration. The extracted spectral values from sentinel 2 and the conventionally determined soil EC were modelled using partial least square regression (PLSR), a method which reduces the variables, used to predict, to a smaller set of predictors. Soil EC is the dependent variables whereas the spectral or reflectance values from multispectral images are the independent variables. The Root Mean Square Error (RMSE), coefficient of determination (R²) and relative percentage deviation (RPD) were used to test the predictive ability of the model. RMSE provides the absolute average error between the measured and the estimated values for samples (Leone *et al.*, 2012). The R² measures the proportion of the total variation accounted for and Table 2.2 shows the prediction of goodness of the model based on this parameter. RPD refers to the ratio of standard deviation (SD) to the RMSE prediction and predictive ability of the model was also based on the relative percentage deviation (Table 2.3).

Table 2.2: Prediction of goodness of the model based on the coefficient of determination (R^2) supplied by Mouazen *et al.* (2010)

R^2	Model performance
0.5 to 0.65	Poor prediction model
0.5 to 0.79	Good
0.80 to 1.00	Excellent

Note. R^2 = coefficient of determination

Table 2.3: Prediction of goodness of the model based on the relative percentage deviation (RPD) supplied by Duran (1983)

RPD	Model performance
RPD <1.0	Very poor model/ prediction
1.0 < RPD >1.4	Poor prediction model
1.4 < RPD > 1.8	Fair model/ Prediction
1.8 < RPD > 2.0	Good model
2.0 < RPD > 2.5	Very good model
RPD > 2.5	Excellent model

Note. RPD = relative percentage deviation;

2.5. RESULTS

2.5.1 Conventional laboratory analysis results for selected soil parameters.

The results obtained reveal that there is a great spatial variation of soil physiochemical properties in the study site. This is crucial especially when developing models and it ensures that the model is not biased. The results obtained shows that the soils at the portion where the soil samples are collected falls under slightly saline soils since it was found to be 1.03 ds/m (Table 2.4). The results show that soil surface salt concentration range from 0.003 to 2.36 ds/m. Therefore, the lower the soil surface salt accumulation, the lower the exchangeable and soluble

Calcium(Ca), the poorer the drainage conditions hence the lower the microbial activity, water holding capacity and soil fertility that results in reduced plant productivity (Sharma *et al.*, 2000). The results show that clay content range from - 3.71 to 20.16, sand percentage range from 0.064 to 90,72 and silt percentage range from 0,064 to 20,92. Based on the results obtained regarding soil particles distributions it shows that there is high variation across the field (Table 2.4).

Table 2.4: Statistical description of soil parameters analysed by conventional laboratory methods (n= 55) in Syferkuil farm, Limpopo, South Africa

Soil parameter Max Min Mean Median SD CV (%)

Soil salinity(EC) 2.36	0.003	1.03	0.35	0.54	52.43
Clay	20.16	3.71	8.18	8.36	5.77	70.54
Silt	20.92	0.06	10.34	11.05	4.42	42.75
Sand	90.72	0.06	80.77	81.28	6.08	7.53
pH(KCl)	8.02	5.23	7.07	7.14	0.53	7.49
pH(H₂O)	8.73	5.9	8.73	8.32	0.51	5.84

Note. SD = standard deviation; max = maximum; min = minimum; CV = coefficient of variation.

2.5.2 Prediction of soil salinity using models developed from multispectral satellite data.

The PLSR model yielded a coefficient of determination (R^2) of 0.468, the Root Mean Square Error (RMSE) of 0.44 ds/m and the RPD of 0.56 ds/m (Table 2.5). It also shows that the coefficient of determination is low. This actively demonstrates that the model performance was unsatisfactory.

Table 2.5: PLSR model performance for predicting soil EC using multispectral satellite in Syferkuil farm, Limpopo, South Africa

Soil parameter R² RMSE RPD Model performance

Soil salinity(EC) 0.468 0.44 0.56 Poor prediction model

Note. R² = Coefficient of determination; RMSE = The Root Mean Square Error; RPD = Relative Percentage Deviation.

2.6. DISCUSSION

2.6.1 Conventional laboratory analysis results for soil surface salt accumulation.

The results obtained demonstrate that there is high spatial variation at the research site because of statistical variables of soil parameters determined. The low soil EC in some portion of study area may possible be result of human activities (i.e., land use management, farm practices). This is because the portion of the farm that was used for this study has been cultivated for the past years and coupled with occasional resting. Conventional methods breakdown soil aggregates hence leading to soil being easily removed or eroded thus affecting soil quality. Furthermore, the natural factors such as the soil type and characteristics of terrain are the main drivers of the spatial variation of soil physiochemical properties. Taking into considerations that at the farm soils are identified as sandy loam texture and the dominant soil forms on the site are Shortlands and Clovelly (Soil Classification Working Group, 1991). Soils that have a higher content of clay conduct more concentration of soil EC than soils that have a higher content of silt and sand particles. Therefore, Shortlands soils do not present acidity issues and are productive. Irrigation and drainage is needed to reduce soil salinity because when irrigating, applying water can help to leach excess salts below the root zone and maintain the desired EC level for the crop growth. However, the application of irrigation systems that contains high amount of magnesium, aluminium and calcium can results in high accumulation of salts in the soil surface (Aldabaa et al., 2015). In addition, at the farm soils are identified as sandy loam texture. Shortlands soils are typically associated with sweet grazing (Fey, 2010). Sandy loam texture is characterized by good drainage, low water holding capacity and high infiltration rate

because of large macro pores.

Another possible reason may be due to climatic conditions (semi-arid for Limpopo, Mankweng area). Taking into consideration that the rainfall ranges from 450 to 630 mm per annum at the study sites. This actively demonstrate that there is high rainfall. In addition, rainfall is more effective in leaching salts or reducing salinity. However, in areas where there is little rainfall, evaporation rates are extremely high and water evaporates rapidly leaving dissolved mineral salts in the upper soil profile, which then build up progressively with space and time. A high concentration of salts at the surface can result in poor soil structure, low fertility and microbial activity and decrease the ability of the soil to support some plants (Sharma *et al.*, 2000).

2.6.2 The performance of model developed from multispectral satellite image (sentinel 2)

The PLSR model yielded a coefficient of determination (R^2) value ranging from 0.19 to 0.48; the relative percentage deviation (RPD) ranging from 0. 67 to 1.29; and the root mean square error ranging from 0.19 to 6.71 of all selected soil properties. The performance of the model was not excellent. Nonetheless, the result shows that approximately half of the soil properties variation could be explained by the reflectance values as recorded in the image. The average deviation of the predicted values from the multispectral images using the PLSR analysis is given by the magnitude of the RMSE value. The low R^2 value indicates that there are confounding factors. The confounding factors might come in the form of noise and errors due to variations in soil surface roughness, geometric and atmospheric effects (Casa et al., 2013). Image noise is any unwanted disturbance in image data that is due to limitations in the sensing, signal digitization, or data recording process. Furthermore, the performance of models might be because of low spectral resolution of the image as compared to hyperspectral data, which have bands with narrower wavelengths (Qi et al., 2017). The results showed low content of soil EC, this has influence on the reflectance spectra, and since soil which has low content of salt normally have high reflectance spectra (Lacerda et al., 2016).

The results obtained in this study, are in line with Mohammad *et al.* (2019) who compared the estimated EC values with ground-truth measurements to evaluate model consistency in Kuh Sefid (Iran). The performance of the PLSR model was

unsatisfactory. Furthermore, the results were supported by Allbed and Kumar (2013) who highlighted that multispectral data has limitations because of the coarse spatial and spectral resolutions, which influences the guality and guantity of the information they provide. The results agree with those obtained by Liao et al. (2013) using spectral reflectance of Landsat ETM image to predict soil texture. The coefficient of determination reported for silt, sand and clay was found to be 0.32, 0.21 and 0.3, respectively. Soil texture and soil EC are related even though soil texture is a physical property and soil EC is a chemical property. This is because soils that have a higher content of clay have higher EC than soils that have a higher content of silt and sand particles. Clay textured soils which have high compaction can increase salinity and decrease the ability of soils to support some plants. Furthermore, Clay textured soils have low reflectance as compared to sandy loam textured soils because of high water holding capacity. This actively demonstrates that as soil moisture increases, spectral reflectance of soil decreases. This means that if there is high amount of water in the soil, the reflectance will go down. Therefore, drier soil such as sandy textured soil reflects more than wet soils (i.e., clay textured soils) (Kumar and Sharma, 2020).

Casa *et al.* (2013) also found that the PLSR model for CHRIS satellite image showed that the model prediction performance for clay, silt and sand estimations was unsatisfactory. Forkuor *et al.* (2017) observed poor estimation accuracy for model performance of the following soil constituents namely sand, silt, clay and CEC when using multispectral satellite image (R^2 of 0.35, 0.54, 0.21 and 0.36; RMSE of 7.57, 5.94, 6.95 and 4.79, respectively). Franceschini *et al.* (2015) also reported poor prediction of K, Ca and Mg using PLSR models developed with spectra data derived from airborne sensor (R^2 of 0.44, 0.52, 0.51; RMSE of 1.28, 7.6, 2.4; RPD of 1.28, 1.47, 1.45, respectively).

The results obtained in this study are however in contrasts with Al-Gaadi *et al.* (2021) who estimated soil salinity using Sentinel 2 satellite images and related conventional laboratory results of soil EC and spectral values using multiple regressions modelling. They found that the generated models show satisfactory results in predicting soil EC. Conforti *et al.* (2015) have reported that the sand percentage was successfully predicted by visible and near infrared spectroscopy with the R² of 0.81 and RMSE of 4.8% for validation dataset. De Santana *et al.* (2018), Curcio *et al.*

(2013), Xu *et al.* (2018), also obtained good R² in their studies (0.70, 0.74, and 0.67 respectively) for predicting the sand percentage of soil from visible and near infrared reflectance spectroscopy using partial least square regression. The difference in the results of these studies may possibly be due to the performance of the models which varies with the study site and the regression techniques used (Gorji *et al.*, 2017; Shahabi *et al.*, 2017). The performance of the models varies with study site because models are calibrated differently based on the soil parameters investigated. For instance, the model that is used in South Africa might not be adopted in some other countries. This is because of different climatic conditions and soil types.

2.7. CONCLUSIONS AND RECOMMENDATIONS

The study revealed the potential of using remote sensing techniques and understanding of soil surface salt accumulation. The results obtained showed that the performance of the model was not excellent or produced low prediction accuracy using multispectral imagery. The use of Sentinel 2 imagery did not give good prediction of all selected soil properties. To improve model performance, future studies should consider an image with better spectral resolution and narrower bands (hyperspectral) to see if there is an improvement in the model performance. The hyperspectral image covers spectral bands narrower than multispectral imagery and image data from several bands are recorded at the same time. Furthermore, hyperspectral image offer much greater spectral resolution than multispectral imagery (i.e., cover two or more spectral bands simultaneously typically from 0.3 m to 14m

CHAPTER THREE

ASSESSMENT OF RILL DEGRADED SOILS IN SYFERKUIL FARM USING REMOTE SENSING TECHNIQUES

ABSTRACT

Assessment and identification of soils degraded by rill erosion is essential for future land use management and planning. Rill erosion is one of the most globally significant environmental soil hazards, which results in severe threat on crop productivity and biodiversity since it can lead to loss of soil quality. The causes of rill erosion can either be natural or human factors. The objective of this study was to identify soils degraded by rill erosion with acceptable accuracy from remote sensing images. The study was conducted at the Syferkuil farm, Limpopo, South Africa. In this study, rill identification and dimensioning was done on a multispectral imagery (Worldview 2 satellite image) using supervised image classification. Quantum Geographic Information System (QGIS) software was used for the visual vectorization of individual rills. The Semi-Automatic Classification Plugin (SCP) in QGIS was used to identify soils degraded by rill erosion in Syferkuil. The classification was done using the spectral angle algorithm. The area was classified into two categories which is rill and no-rill. Raster object was converted to vector object using polygon trace tool for further object-based processing. For the purpose of assessing the accuracy of the supervised image classification, reference data was created, and the error matrix was calculated, and the results indicated the user's accuracy and producer's accuracy for no rill erosion and rill class were obtained.

The results of the study were not satisfactory using the supervised classification of the Worldview 2 satellite image. The possibility therefore exists that some soil erosion features may have been classified as non-erosion features. In addition, the low separability of classes limits the applicability of the supervised classification methods, particularly in spectrally complex erosion areas. The overall classification performance or accuracy of 47.91% with a Kappa coefficient of 0.41 was obtainable. Therefore, these remote sensing techniques, although there is much room for improvement, can contribute into identifying and quantifying soils degraded by rill erosion especially in data scarce environment and resources constrained province (i.e., Limpopo, Mankweng Area).

Keywords: Soil degradation, Rill erosion, Syferkuil, Remote sensing, Multispectral image (Worldview 2 satellite image), Supervised mage classification.

3.1 BACKGROUND

Monitoring soil quality degradation is essential towards land use management studies and reclamation programmes or rehabilitation strategies. The first step towards rehabilitation and reclamation of different soils is by providing accurate soil information and having an understanding of soil degradation forms. This will help in identifying high risk areas for future land use management and planning. This study will only focus on soils degraded by rill erosion. Rill erosion is a process of soil degradation that removes soil materials from one point on the earth to be deposited elsewhere through processes such as detachment, suspension, transportation and, mass movement (Vanmaercke *et al.*, 2021; Li *et al.*, 2016) .It is an important form of soil erosion that contributes greatly to soil degradation and loss in South Africa. Monitoring soil quality degradation can be precisely achieved through the usage of remote sensing as an alternative approach due to its advantages over conventional methods (i.e., Field surveys). Field surveys delay the process of acquiring necessary soil information of high accuracy in a short period of time and are limited to small areas (Rossel *et al.*, 2011; Odindi *et al.*, 2017).

In remote sensing (RS), data is collected using either passive or active remote sensing techniques, without making physical contact with the object itself (Chauhan, 2015). Active remote sensing, capture EM radiation in the visible spectrum and has its own source of light or emits its own energy. However, in passive remote sensing the source of signal is the sun, which emits EM at its highest intensity between the ultraviolet and infrared. Remote sensing has the capability to identify soils degraded by rill erosion with acceptable level of details (Kumar, 2013; Morshed *et al.*, 2016; Taghadosi *et al.*, 2018). Remote sensing is able to cover a large area of land especially when data is required for large areas. Moreover, remote sensing is faster, inexpensive, non-destructive, facilitate long term monitoring, and accurate monitoring tool. In addition, remote sensing allows for past, present, and near real time monitoring of objects of interests (Chauhan, 2015).
The multispectral satellite imagery has been widely studied in previous research and has been found to be a very promising tool for assessment rill degraded soils (Karydas *et al.*, 2020; Desprats *et al.*, 2013). The multispectral satellite imagery has the capability to provide improved spatial resolution and image acquisition is not costly which makes it effective in terms of classifying, analysing, monitoring and mapping soil degradation forms at different locations. RS images can provide detailed but much reduced version of reality or 3D view of object of interests. This can cover different areas which might be difficult to cover using some other terrestrial means (Karydas *et al.*, 2020; Desprats *et al.*, 2013).

To contribute to the valuable information and better understanding on the effects of soil degradation caused by rill erosion with the use of remote sensing, our study was focused on the Mankweng area (Syferkuil farm) in Limpopo province. This study will establish if rill degraded soils could be identified and quantified using remote sensing. The objective of this study was to identify soils degraded by rill erosion with acceptable accuracy from remote sensing images.

3.2 PROBLEM STATEMENT

Rill identification and assessment of intensity using field work can provide accurate results and have been preferred by most researchers, but they are only applicable to small areas. The main difficulties lie on meeting high demands of detailed soil information in short period of time with reasonable cost (Steinberg *et al.*, 2016). Field surveys are time consuming and labour intensive particularly for regional level in which data is required for large scale applications. Moreover, these traditional methods (i.e., field work or field surveys) are difficult to replicate and limited to small areas (Odindi *et al.*, 2017). Therefore, with the current demand for up to date and accurate soil information on soil degradation, these methods are ineffective. Moreover, they delay the process of acquiring necessary soil information of high accuracy in a short period of time (Rossel *et al.*, 2011).

The challenges of conventional methods, however, could be addressed by the use of remote sensing techniques. Many researchers have identified remote sensing as an alternative method for assessment and identification of soils degraded by rill erosion due to its advantages. This is because remote sensing techniques can provide rapid analysis of soil information, cover large land surface and facilitate long term monitoring at an acceptable level of detail (Chauhan, 2015). A multispectral imagery is able to provide excellent spatial coverage of a large area and making it easy to identify soils degraded by rill erosion (Gorji *et al.*, 2017). Remote sensing (RS) images can provide detailed, but much reduced version of reality or 3D view of object of interests. These imageries can provide valuable information that might not be able to obtain using other methods since our visual perception is limited to some portions of electromagnetic spectrum (EMS). This means that it can provide the records of soil degradation forms at those specific times the images were captured and cover different areas which might be difficult to cover using field surveys. This can help to observe change that might be occurring over time and to understand the spatial extent and rate of this problem.

3.3 LITERATURE REVIEW

3.3.1 Work done on the problem statement

3.3.1.1 The importance of identifying soils degraded by rill erosion using remote sensing.

Soil erosion globally is an intense, poorly controlled process and there is lack of upto-date soil information. Soil erosion is a serious problem in the entire world and a major threat of land degradation in South Africa particularly in Limpopo (Rahmati *et al.*, 2016; Le Roux and Sumner, 2012). It is also considered as one of the most critical environmental issues due to the devastating impacts on agricultural lands. Therefore, soil erosion is a dynamic process requiring constant monitoring while keeping up-todate information on its spatial distribution. Soil conservation and rehabilitation measures and understanding the dynamics of soil degradation and driving factors is a crucial step (Nwilo *et al.*, 2021; Le roux and Sumner, 2012). It is important for modelling erosion hazard of the area or high-risk area for future land uses management and planning (Ogbonna, 2012).

Rill erosion can lead to land abandonment and threaten food security since soil fertility or nutrient status of the soil might be compromised. In turn, many farmers abandoning their farmlands due to the threats posed by rill erosion and then cause economic effects on farming communities. In addition, rill erosion can results in decline soil fertility, poor soil structure, poor drainage conditions. This is because

during heavy rainfall essential nutrients are leached away, and as a results leading to decreased soil quality and capability to sustain crops (Dewitte *et al.*, 2015).

3. 3.1.2 Rill degraded soils monitoring, identification and assessment using remote sensing.

Various studies have been done for rill identification and assessment of intensity using remote sensing. Saadat *et al.* (2014) conducted a study of mapping rill erosion using Landsat in Iran. The results obtained showed that the proposed method is able to produce a rill erosion intensity map with an accuracy of 96% at this study location.

Desprats *et al.* (2013) conducted a study of mapping rill erosion using Quick Bird and SPOT satellite imagery in Tunisia. They found that the high-resolution imagery (Quick Bird) is a valuable tool from which one can extract the consequences of soil degraded by rill erosion whereas it remained fairly insignificant with the SPOT type. However, this methodology used demonstrates the potential for extracting rill erosion features from the imagery, but more importantly it discusses a GIS analysis that can identify elements with soil erosion traits from among all the linear features.

Fiorucci *et al.* (2015) conducted a study of mapping and measuring rill erosion using GeoEye -1 panchromatic stereo images in Italy. In this study, they found that the proposed method is faster than field work, improves the ability to map these features over large areas, which are applicable to detailed scales and analyses, and other more traditional method. Karydas *et al.* (2020) conducted a study of mapping rill erosion in which data from sentinel 2 images was used. They have found that the multispectral imagery (sentinel 2) is suitable for future erosion assessments with G2 model. Basically, G2 is quantitative algorithm for mapping soil loss and sediment yield rates on month-time intervals. G2 model proved to work as a rapid and at the same time flexible mapping tool.

Gafurov (2022) conducted a study using the trained rill erosion convolutional neural network (RECNN) for automated rill erosion detection from remote sensing data in Russia. The results of this study showed accuracy level of 0.62, F1-measure was 0.76, and loss-function was 0.27. Furthermore, it was found that not a single case of detection of gullies or ground roads, which are abundant in the study area, instead of rills, was recorded.

3.3.1.3 The use of supervised image classification for identification and assessment of rill degraded soils.

In supervised image classification, there is no limited control over identity of classes. Training area is referred as an area of known identity delineated on the digital image within its coordinate system. The key characteristics of training areas are number of pixels, size of the training area, shape of the training area, location of the training area, and number of training area (Eastman, 2003). However, the training data not often defined spectrally, and the training areas must be selected carefully to minimize errors. Moreover, not all spectral classes may be known to the users. In most instances, the decision to utilize this image classification depends on the study area, the skills of individual processing the image, and the spectral distinctness of the classes. The spectral class are inherent in the multispectral imagery or remotely sensed data whereas the information class are defined by human beings (Eastman, 2003).

3.3.1.4 Factors affecting the spectral signatures of rill degraded soils.

According to Taruvinga, (2008), the spectral signature of soil erosion forms differs. In addition, the spectral signature of soils depends on the moisture content, organic matter content, texture, structure and iron oxide content (Aggarwal, 2004). Spectral characteristics of vegetation differ with wavelength and the pigment leaves of plant reflect green wavelengths and strongly absorb red and blue wavelengths. Bare soil and vegetation have different spectral characteristics and are completely different and need to be dealt with separately when selecting training areas. The heterogeneous nature of soil degraded by rill erosion makes it complex to differentiate with the surroundings, thereby posing a challenge to the classification technique (Aggarwal, 2004).

3.3.1.5 The possible human and environmental factors that influence rill erosion initiation and development.

According to Chaplot *et al.* (2013) and Mararakanye and Summer, (2017), Soil erosion forms are influenced by soil type, bedrock lithology and structure, precipitation, slope angle, vegetation and land use. Although, rill erosion is a natural process, it is accelerated by human activities and rainfall or climatic conditions (Vanmaercke *et al.*, 2021; Li *et al.*, 2016).

Clay textured soil is more compacted and has higher structural stability that resists soil erosion whereas sandy or silty textured soils are less susceptible to soil erosion because of larger pores and high infiltration rate which results in high leaching of soil materials (Gandariasbeitia et al., 2017). Overgrazing can reduce ground cover and enable erosion by wind and rain. Therefore, vegetation plays a crucial role in reducing rill erosion because plants roots bind soil particles together and increase structural stability of soils that are not easily leached out or eroded by soil erosion agents (i.e., water or wind). Therefore, the more water flowing over the surface or land, the more soil particles are leached out or transported from one place to another. Meaning that farmland that has no vegetation is vulnerable to rill erosion as compared to the one having vegetation cover. Land use practices can influence rill erosion. For instance, the conversion of natural ecosystem to pasture land can lead to high rates of rill erosion and loss of top soil and nutrients. The slope can have a major influence on rill erosion. Meaning that when the slope is longer (length), surface area of water collection increases and therefore increase the water surface runoff (Le Roux and Sumner, 2012).

3.3.2 Work not done on the problem statement.

The study is done at a farm level requiring much detail with a potential to apply on a large scale. The remote sensing technique used can vary with the location or study area and the feature being observed. Most of the studies have focused on the soil erosion origin and contributing factors. Moreover, there is little work done that utilizes remote sensing for assessing soil degradation on South African soils and particularly in Limpopo. Therefore, this study will establish if soils degraded by rill erosion could be identified and quantified using remote sensing as an alternative method to the traditional methods (field work or surveys).

3.4 RESEARCH METHODOLOGY

3.4.1 Description of the study area

The study was conducted in Syferkuil, experimental farm of the University of Limpopo (Figure 3.1). The farm is about 1,650 ha in size and 13 km from the main campus of the University. The climate is semi-arid with rainfall of relatively 450-630 mm per annum, experiencing about 400 mm of summer rainfall. The farm is exposed to temperatures ranging between 14°C during winter periods and 35°C in summer. The soil forms are Hutton and Glenrosa and composed of seven minerals which include plagioclase, K-feldspar, amphibole, quartz, interstratified illite or smectite, talc and kaolinite dominated by quartz and interstratified illite or smectite (Molepo *et al.,* 2017). The soils from Syferkuil farm is moderately shallow to deep and consists of the following texture classes; sandy loam, loamy sand and sandy clay loam (Phefadu and Kutu, 2016).



Figure 3.1: Study site in the Syferkuil farm, Limpopo

3.4.2 Data collection

3.4.2.1 Remote sensing

a) Identification of soils degraded by rill erosion using supervised image classification

In this study, the Quantum Geographic Information Science (QGIS) software was used for the visual vectorization of individual rill erosion forms. The Semi-Automatic Classification Plugin (SCP) in QGIS was used for identification or assessment of soils degraded by rill erosion. Supervised image classification was done using QGIS to identify soils degraded by rill erosion in Syferkuil. The classification was done using the spectral angle algorithm. The portion of the farm that was used for this study has an area of 61 ha. The area was classified into two categories which is rill and no rill. Raster object was converted to vector object using polygon trace tool for further object-based processing. The SCP is defined as a free open source plugin for QGIS (Congedo, 2016) that allows for the semi-automatic classification of remote sensing images for the visual vectorization of individual rills. Supervised image classification involves collecting data from the training area followed by a classification step or stage then output stage (Figure 3.2). Rill identification and assessment was done on multispectral satellite imagery [Worldview 2 satellite image] as a form of feature extraction. This image multispectral imagery had one panchromatic band with 46 cm spatial resolution and eight bands with 1.85 m spatial resolution. This satellite image had wavelengths covering from 400 nm to 1040 nm. The supervised remote sensing image classification approach was used based on only two categories which are rill and no rill. It is the digital image processing that commonly group pixels to represent land cover attributes. The sample size was selected in an image that represents the specific classes and then directed the image processing software to use these training sites as references for the classification of all other pixels in the image. It used the spectral information represented by the digital numbers in many spectral bands and attempt to classify each individual pixel based on this information. The computer used an algorithm in order to determine spectral signatures for each training class and then compares each pixel in the image to these signatures and labels it as the class it most closely resembles digitally. The training areas were selected based on what it is known on the ground then digitizes a polygon within that particular area. The spectral signature of each pixel was matched with the training signatures and the image was classified accordingly. Pixels which have the same spectral characteristics were identified as belonging to the same class and assigned or given unique number or colour. Pixels are defined as the small representation of reality or objects of interest.



Figure 3.2: Flow charts showing supervised image classification.

3.4.3 Data analysis

The multispectral imagery (Worldview 2 imagery) was used to discriminate rill and no rill erosion in Syferkuil farm. The SCP in QGIS was used to classify the image or for the visual vectorization of individual rills to generate statistical results of the map accuracy. SCP has the ability to discriminate land cover feature and offers a potential for mapping individual rills.

3.4.3.1 Accuracy assessment

Accuracy assessment and check was done by generating validation points. The error matrix is the method used for assessing the degree of accuracy (Mather, 2004) and has been widely used in classification accuracy assessment. Error matrix is a square which contains rows and columns that are equal to the number of categories whose classification accuracy is being assessed (Lillesand *et al.*, 2008). The results of the error matrix were interpreted using the producer's accuracy, user's accuracy, overall classification accuracy and the Kappa classification. The overall classification accuracy measures the producer's accuracy as well as the user's accuracy. The user's accuracy measures the errors of commission while the producer's accuracy measures the errors of omission. The Kappa coefficient is the difference between the actual agreement in the error matrix and the agreement occurring by chance (Persello and Bruzzone, 2010).

RESULTS AND DISCUSSION

3.5 RESULTS

The results of the digital image processing (supervised image classification) of the area classified into two categories which is rill and no rill erosion showed a low overall accuracy of up to 47.91%. The accuracy results produced is very low and the kappa statistic of 0.41 indicates none to slight agreement.

3.5.1 Supervised image classification results

The map shown in Figure 3.3 illustrates the cultivated area with the selected training areas created by using Region of interest (ROI) polygon.



Figure 3.3: The cultivated area with the selected training areas

Figure 3.4 below illustrates the validation points selected randomly to provide information and evidence that the classification produced the expected results. The random selection was done automatically by SCP and this map shows that it covers the entire study area.



Figure 3.4: Randomly selected validation points

Figure 3.5 below shows the classification done with spectral angle algorithm. The classification is showing the rill and no-rill regions. It shows that areas affected by rill are identified positively. Areas like inter-plot paths are also identified as rill regions. Furthermore, some whole plots are also depicted as representing rills. The veracity of the model prediction is clearly conveyed by the accuracy results. This means that the predictive ability of the model depends on the accuracy of the results. The higher accuracy means better performance.



Figure 3.5: Classification done with spectral angle algorithm.

3.5.2 Accuracy results

The supervised image classification was used for classification of soils degraded by rill erosion. The Kappa statistic of 0.41 indicates moderate agreement (Table 3.1). The overall classification accuracy calculated of 47.91% indicates that only 47.91% accuracy was classified correctly (Table 3.2). The user's and producer's accuracy indicate the performance of supervised image classification for rill erosion and no rill erosion identification. However, a 47.23% of producer's accuracy in rill class indicates more errors of omission while 45.80% of producer's accuracy in no rill erosion class. Kappa is the most used indicator of classification accuracy and is used to assess the agreement between the dependent variable and independent variable. In addition, Kappa coefficient can also be used to assess the performance of a classification model (Persello and Bruzzone, 2010). A Kappa value of 0.41 in rill class is problematic because the classification accuracy indicates moderate agreement rill and no rill erosion. The accuracy results produced is very low as indicated in Table 3.3.

Kappa coefficient	Strength of agreement
≤ 0	No agreement
0.01 - 0.20	None to slight agreement
0.21 - 0. 40	Fair agreement
0.41 - 0.60	Moderate agreement
0.61 - 0.80	Substantial agreement
0.81 - 1.00	Almost perfect agreement

Table 3.1: Showing the classification to interpret the strength of agreement based on the Cohen's Kappa value (Altman, 1997; Landis, 1977)

Table 3.2: Showing area-based error matrix, reference, standard error, confidence interval, producers' accuracy, users' accuracy and kappa classification values.

>AREA BASED ERROR MATRIX

> REFERENCES

V_ classified	0	1(rill)	2(no rill)	Area	Wi	
0	0.00	0.00	0.00	2529015.50	0.69	
1	0.00	0.012	0.11	440931.25	0.12	
2	0.00	0.019	0.17	712232.00	0.19	
Total	0.00	0.031	0.28	3682178.7	5	
Area (metre ²)	0.00 (0.0	0.28	3682178.7	5	
SE	0.00	0.02	0.02			
SE area	0	83767	83769			
95% Cl area	0	164184	164184			
PA[%] nan		47.24	45.76			
UA [%] nan		49.40	50.03			
Kappa coeffi	cient	0.41	0.41			

Note. PA = producer's accuracy, UA = user's accuracy, SE= standard error and, CL = confidence interval.

Overall accuracy [%] = 47.91

Kappa coefficient = 0.41

Two categories	Producer's accuracy (%)	User's accuracy (UA)	Overall Accuracy (%)	
Rill erosion	47.24	49.40	48.31	
No rill erosion	45.76	50.03	47.52	
Overall Accuracy	(%) 47.91			

Table 3.3: Worldview 2 satellite image accuracies (%) for rill erosion and no rill erosion.

3.6 DISCUSSION

The results obtained reveal that the digital image processing (supervised image classification) of the area classified into two categories which is rill and no rill erosion showed a low overall accuracy of 47.91% and the total Kappa coefficient of 0.41. The accuracy results produced is very low and the Kappa statistic of 0.41 indicates moderate agreement between rill and no rill erosion. Therefore, the results were not satisfactory using multispectral satellite image [Worldview 2 satellite image]. This may be due to limited spectral bands, atmospheric conditions and soil properties (Shruthi *et al.*, 2011; Taruringa, 2008). The heterogeneous nature of soil degraded by rill erosion makes it complex to differentiate between soil erosion feature and non-erosion features, thereby posing a challenge to the classification technique. The error matrix (pixel count values), Reference, Classified and Pixel Sum values are in Appendixes 1 and 2.

In a previous study by Torkashvand and Alipour (2009) using supervised image classification for assessment of rills using remote sensing techniques in plain physiography of Iran, the researchers found out that the accuracy decreases where there are other land uses such as cultivation due to similarity in spectral characteristics. Thus, it is important to classify where there is bare soil in order to increase the accuracy. Furthermore, within the study area, the spectral reflectivity of

soil erosion varies considerably, and in some cases tends to be similar to nonerosion features (for example bare soil). The possibility therefore exists that some soil erosion features may have been classified as non-erosion features. Similarly, Sepuru and Dube (2017) emphasised that the low separability of classes limits the applicability of the supervised classification methods, particularly in spectrally complex erosion areas.

Another possible reason is that the supervised image classification depends on the training sites, the skill of the person processing or classifying an image and the spectral distinctness of the classes. This method of classification requires close attention to the development of training data and if the data is poor and not representative, the classification results will also be poor decreasing the accuracy of results. However, the training data not often defined spectrally, and the training areas must be selected carefully to minimize errors. Moreover, not all spectral classes may be known to the users. The spectral class are inherent in the multispectral imagery or remotely sensed data whereas the information class are defined by human beings.

These results are however in contrast with those obtained by Floras and Sgouras. (1999) who found high overall accuracy of 83.94% using supervised image classification method (The Gaussian maximum likelihood classifier). This is because their classification was assisted by Digital Elevation Model (DEM). Moreover, they used the Landsat 5 images while we used the worldview imagery. Landsat 5 image is easy to obtain and has a wide coverage and, is suitable for the large-scale land cover studies (Wang *et al.*, 2019)

This is supported by various studies (Phinzi and Ngetar (2017); Munyati and Ratshibvumo (2011); Singh *et al.* (2015) using Landsat 5 in assessing rills erosion. The results of these studies indicated that an overall classification performance or accuracy of above 81% was obtainable. The results are in line with those obtained by Sepuru and Dube (2018) in mapping spatial distribution of three eroded area using Sentinel 2, achieving an overall classification accuracy of 81,90%. In another study by Azad (2019), the author found a high overall accuracy of surface erosion classification of 69% using the supervised image classification. Therefore, higher spatial resolution allows finer grain details to be discerned in the imagery. Satellite

imagery with relatively higher spatial resolution allows the classification technique to detect the smaller rills which were omitted from low spatial resolution imagery. The image acquisition is costly and limited due to their small swath width (Seutloali *et al.*, 2016, Shruthi *et al.*, 2011; Ranga *et al.*, 2015; Le Roux and Marakanye, 2012).

3.7 CONCLUSION AND RECOMMENDATIIONS

This study focused on identifying soils degraded by rill erosion using remote sensing techniques. In this study, the results were not satisfactory from multispectral satellite image [Worldview 2 satellite image] using digital image processing called supervised image classification. Despite low prediction accuracy in this study, remote sensing is a promising tool since it can provide information beyond our human visual perception. Furthermore, remote sensing can provide the record of soil degradation form at those specific times the images were captured and cover different areas which might be difficult to cover using field based methods.

Even though the results were not satisfactory, remote sensing can serve an alternative method to field based methods because the results of this study can be greatly improved through the use of much higher spatial resolution imagery and narrow bands (hyperspectral satellite image) to see if there is an improvement in the accuracy level. This is because satellite imagery with relatively higher spatial and spectral resolution (i.e., hyperspectral imagery) allows the classification technique to detect the smaller rill erosion features which were omitted from low spatial resolution imagery.

CHAPTER FOUR

ASSESSING SOIL ORGANIC CARBON STATUS USING MULTISPECTRAL IMAGES IN SYFERKUIL FARM

ABSTRACT

Soil organic carbon (SOC) depletion which is directly linked to human and natural activities poses a major threat to agricultural productivity since it can lead to reduced soil and water quality, soil fertility and nutrient status. Assessing and monitoring soil quality degradation is vital in terms of practicing precision agriculture and future land use management and planning. The main objective of this study was to determine SOC status with acceptable accuracy from a remotely sensed image. The size field area of the plot sampled is 61 ha. Ninety seven soil samples were collected and a regular sampling grid strategy of 50 m by 100 m was followed in Syferkuil Farm, South Africa, Limpopo. Global positioning system (GPS) was used to record the exact location of each sampled point in each grid. In this study, the SOC of the soil samples was determined conventionally in the laboratory using a Walkley Black method. The coordinates of the sampling points were used to extract spectral value from the multispectral image (Sentinel 2) using QGIS. The spectral values and the conventionally determined SOC were modelled using partial least square regression (PLSR). The results showed low prediction accuracy of SOC with R² of 0.41, Root Mean Square Error (RMSE) of 0.53%, and the relative percentage deviation (RPD) of 1.21 using the PLSR model developed from the multispectral image (Sentinel 2). The result shows that approximately half of the SOC variation could be explained by the reflectance values as recorded in the image. Despite low prediction accuracy, remote sensing techniques are potential tools to monitor and detect SOC unlike in the old days, when conventionally laboratory methods were the only means of assessing soil organic carbon depletion or status. Sentinel 2 could be used to make preliminary study of SOC before detailed in situ assessment could be done. Thus, it is recommended that an image with better spectral resolution (hyperspectral) be investigated to see if there be an improvement in the model performance.

Keywords: Soil degradation, Soil organic carbon, Remote sensing, Multispectral image (Sentinel 2 imagery), PLSR, Syferkuil farm

4.1 BACKGROUND

Soil degradation has become the most serious global environmental issue that needs a serious attention due to its devastating impacts on agricultural productivity. This can lead to reduced water quality, soil fertility and nutrient status especially in arid and semi-arid regions (Shi et al., 2019). Thus, it is of crucial importance to monitor and assess soil quality degradation for sustainable and precision agriculture. This can help with reclamation and rehabilitation of soils and minimizing the risks of soil degradation forms in different regions. In this study, one form of soil quality degradation to monitor or assess is soil organic carbon (SOC) depletion. Assessing SOC status is vital in terms of soil health monitoring and environmental management. SOC depletion is directly linked to human and natural factors (FAO of United Nations, 2012). In agricultural land, SOC is a key indicator of soil fertility due to its beneficial effects on soil properties. This is because SOC has potential to increase soil fertility, quality, and cation exchange capacity and enhances the water holding capacity (Mccauley et al., 2017; Zhu et al., 2018; Bangroo et al., 2020). Moreover, SOC can contribute to high stable aggregate structures, that are resistant to soil degradation forms hence binds soil particles together. Soil carbon contains a major proportion of carbon that is considered to be three times larger than in the atmosphere and terrestrial vegetation (Houghton, 2007).

Over the past decades, conventional laboratory methods were the only means of assessing SOC depletion. These methods delay the process of acquiring necessary soil information of high accuracy in a short period of time, due to their long procedure (Rossel *et al.*, 2011). Moreover, they are difficult to replicate, time consuming, labour intensive, and are limited to small localized scales (Odindi *et al.*, 2016, Angelopoulou *et al.*, 2019). Remote sensing techniques have the potential to serve as an alternative approach due to its advantages over conventional laboratory methods (Kumar *et al.*, 2016). In previous research, remote sensing have been widely used and found to be a promising tool for assessment of soil degradation forms. These include simultaneous collection of data systematically, inexpensive, non-destructive, facilitate long term monitoring and non-requirement of chemical reagents (Chauhan, 2015; Angelopoulou *et al.*, 2019). In addition, it can provide rapid analysis of soil information and covering large land surface at an acceptable level of detail. In this study, the multispectral imagery (i.e. Sentinel 2 satellite imagery) was

used to extract spectral values using the coordinates of the sampling points in QGIS. The multispectral image has the capability to provide improved spatial resolution and excellent perspectives on land related studies and have the capability to produce significant results in SOC estimation and soil health monitoring (Wang *et al.*, 2018; Odindi *et al.*, 2016; and Zhou *et al.*, 2020). Therefore, the purpose of this study is to determine SOC status with acceptable accuracy from remote sensing. The study was focused on the Syferkuil farm in the Limpopo province.

4.2 PROBLEM STATEMENT

Soil carbon is a key indicator of soil fertility or quality since it can influence the three categories of properties (physical, chemical and biological) (Houghton, 2007). For instance, water holding capacity, microbial activity and cation exchange capacity of soils (Mccauley *et al.*, 2017; Zhu *et al.*, 2018; Bangroo *et al.*, 2020). In addition, it can contribute to high stable aggregate structures, that are resistant to soil degradation forms. This is because SOC has the capability to bind soil particles together.

Since there is higher demand of detailed soil carbon information, remote sensing techniques have the ability to provide soil carbon information in short period of time with reasonable cost. Remote sensing (RS) could be used as an alternative method to the conventional laboratory methods due to its advantages. It can provide rapid analysis of soil information and covering large land surface at an acceptable level of details (Chauhan, 2015). It has the potential to provide valuable information that might not be able to obtain using other methods since our visual perception is limited to some portions of EMS (Electromagnetic spectrum). This means that it can provide information beyond our human visual perception. Furthermore, it can provide the records of soil degradation forms at those specific times the images were captured and cover different areas which might be difficult to cover using some other terrestrial means. This can help to observe change that might be occurring over time and to understand the spatial extent and rate of this problem. RS images can provide detailed but much reduced version of reality or 3D view of object of interests (Gorji *et al.*, 2017).

Multispectral imagery is able to provide excellent spatial coverage of a large area, and making it easy to obtain soil carbon information (Gorji *et al.*, 2017). Despite the wide use of conventional laboratory methods, these methods are time consuming,

expensive, and labour intensive especially when data is required for large farmlands (Rossel *et al.*, 2011). Therefore, with the current demand for up to date soil carbon information, conventional methods are ineffective.

4.3 LITERATURE REVIEW

4.3.1 Work done on the problem statement

4.3.1.1 The importance of assessing soil organic carbon status.

Soil organic carbon depletion as a result of natural and human activities can lead to reduced crop productivity due to lower moisture retention and nutrient status (FAO of United Nations, 2012). This is because, high level of SOC depletion influences water retention properties and cation exchange capacity (Franzluebbers, 2002). Soil organic matter (SOM) plays a crucial role in physical, chemical and biological functions of agricultural soils (Houghton, 2007). SOC is a key indicator of soil fertility due to its beneficial effects on soil properties and contributes to high stable aggregate structures, that are resistant to soil degradation forms (i.e. soil erosion) because SOM binds soil particles together. In addition, aggregates stability tends to determine soil erodibility and influence water infiltration since it gives the best prediction of erosion.

4.3.1.2 The use of remote sensing in assessing, monitoring, and mapping soil organic carbon depletion.

A study was conducted by Mallik *et al.* (2022) in India to map and predict soil organic carbon (SOC) using remote sensing and terrain data. The results showed that the mean value of SOC status was 0.77% with value ranging from 0.043% to 2.87%. The results indicated that the EBKR (Empirical Bayesian Kriging Regression) model is unbiased and accurate. In this work, the root mean square error (RMSE) and the coefficient of determination (R^2) which were used for validation of model outputs was found to be 0.094 and 0.936 respectively. In addition, they found that the composite, band 6 (vegetation red edge) of sentinel 2, slope and elevation gives the best prediction for SOC.

This was supported by Suleymanov *et al.* (2021) who estimated and mapped the spatial distribution of organic carbon using remote sensing. In this work, they used the Sentinel-2A satellite data and the linear regression method. The results show

that the coefficient of determination (R²), RMSE, and the RPD was found to be 0.58, 0.56, and 1.61 respectively. The linear regression model performance for SOC prediction was found to be satisfactory. This was supported by various studies using Sentinel 2 (e.g. a study by Vaudour *et al.* 2019 in France with the RPD of 1.51 and a study by Castaldi *et al.* (2019) in Germany, Belgium, and Luxembourg in evaluating the capability of the Sentinel 2 data for SOC prediction based on PLSR and RF models. The ratio of performance to deviation (RPD) was higher than 2 in Luxembourg (2.6) and German (2.2) site, while it was 1.1 in the Belgian area. According to Castaldi *et al.* (2019) the prediction accuracy obtained by Sentinel 2 data is generally slightly lower than that retrieved by airborne hyperspectral data. According to Shen *et al.* (2015) the PLSR model achieves better accuracy within the laboratory spectral data than in the field data.

Nabiollah *et al.* (2019) conducted a study in Iran in assessing SOC under land-use change using random forest models. In this study, spectral data was derived from Landsat imagery. In their findings, Nabiollah *et al.* (2019) found that the accuracy was good with RMSE of 3.53 and coefficient of determination of 0.67.

4.3.1.3 The spectral bands to predict soil organic carbon

Wang *et al.* (2010) reported 440, 560, 625, 740, and 1336 nm as the principal spectral bands to predict SOC. Nocita *et al.* (2014) suggested that the spectral region between 580 and 680 nm was sufficient to predict SOC. Bangelesa *et al.* (2020) found that that important wavelengths at around 2000–2200 and 1400–1500 nm are key to predict SOC and The best model results were obtained with transformed spectral data, with the key wavelengths to predict SOC values mostly localised around the visible range (400–700 nm).

4.3.1.4 The limitations of remote sensing techniques in soil organic carbon assessment

The main drawbacks of remote sensing techniques include low signal to noise ratio due to a short integration time over target area and the atmospheric absorptions interfering with the spectral measurements. Furthermore, mixed pixels contains more than bare soil surface and there is a need for geometric and atmospheric corrections (Minu *et al.*, 2017). Remote sensing techniques are mainly affected by external factors such as moisture, surface roughness, vegetation cover, structure,

and change in atmospheric conditions. Soil moisture content tends to influence the prediction accuracy of different sensors. This is because as soil moisture increases, the reflectance of soil decreases. This actively demonstrates that soil moisture is inversely proportional to the reflectance. For instance, if there is high amount of water in the soil, the reflectance will go down. However, drier soil such as sandy textured soils reflects more than wet soils .The surface roughness is also considered as one of the main factors that influence the prediction accuracy. For instance, the smaller local surface roughness, the greater the spectral reflectance (Ben-Dor, 2002; Guerschman *et al.*, 2009).

4.3.1.5 Human and environmental parameters related to soil organic carbon depletion.

The natural factors (i.e., slope, rainfall and soil type) and human factors (i.e., tillage practices) leads to the development and initiation of soil degradation forms and results in decline in SOC status (Phinzi *et al.*, 2020 ; Mohammed *et al.*, 2020). This means that the upper portions of the topsoil which composed of high concentration of organic materials can be easily removed and results in decline in SOC status. Furthermore, low areas have high SOC status due to high concentration of clay content as compared to high areas with low clay content and high erosion rates. The above mentioned natural factors can limit the soil ability to provide essential nutrients that play a crucial role in plant productivity and growth. These natural factors tend to vary with location and farm management. Furthermore, soil organic costs for soil reclamation and rehabilitation (Phinzi *et al.*, 2020; Mohammed *et al.*, 2020).

4.3.2 Work not done on the problem statement.

In the available literature, many researchers have used remote sensing in monitoring soil degradation in different regions that differ with climate, soil, and land use management. It is used as an alternative method to the conventional laboratory methods due to its advantages. Despite various studies conducted for assessing SOC status with the use of remote sensing techniques. There is little work done that utilizes remote sensing and models for assessing soil degradation forms on South African soils and particularly in Limpopo. Most of the studies have focused on the main factors that influence soil degradation rather than developing models that will assist in soil health monitoring and land use management studies. However, the performance of the models varies with the location or study area and the feature being observed. The objective of this study is to determine SOC status with acceptable accuracy from remote sensing.

4.4 RESEARCH METHODOLOGY

4.4.1 Description of the study area

The study was conducted in Syferkuil, experimental farm of the University of Limpopo (23°51'0" S and 29°42'0" E) with an elevation of 1.325 meters (Figure 4.1). The farm is about 1.650 ha in size and 13 km from the main campus of the University. The research site on which this study focused on had an area of 62 ha, a part of the farm leased to ZZ2. The study site falls under semi-arid climate consisting of annual rainfall varying from 450 mm to 630 mm most of which is received from November to March (Molepo *et al.*, 2017). The soil at the farm is identified as sandy loam texture and the dominant soil forms on the site are Shortlands and Clovelly (Soil Classification Working Group, 1991). The portion of the farm that was used for this study has an area of 61 ha, a part of the farm leased to ZZ2. The research site has been cultivated since from 2011 coupled with occasional resting.



Figure 4.1: A map of Syferkuil farm, Limpopo, South Africa.

4.4.2 Data collection

- 4.4.2.1 Conventional methods
 - a) Field sampling

A grid sampling of 50 m by 100 m was followed to collect 98 soil samples. A shovel was used to take soil samples from the top 5 cm of soil surface. Then samples were placed in plastic bags and were taken to soil laboratory. Global positioning system (GPS) was used to record the exact location of each sampled point in each grid. This was done to be able to locate each sample point.

b) Soil organic carbon (SOC) analysis

Upon the arrival at the soil science laboratory of the university of Limpopo, the gravel fraction was removed, and the samples were left to dry for about a week. After drying, the soil samples were sieved using a 2 mm sieve for soil organic carbon determination. The soil organic carbon of the soil samples was determined conventionally in the laboratory using the Walkley-Black method (Walkley and Black, 1934). This method provides precise and accurate results for soil organic carbon (SOC) determination. Table 4.1 below shows the classification of soil based on the soil organic carbon levels. Low SOC level indicates lower moisture retention and nutrient status (FAO of United Nations, 2012) whereas high level of SOC influences soil properties which results high soil quality or fertility.

SOC (%)	SOC levels	
0.00 - 0.5	Extreme low	
0.5 – 1.0	Low	
1.0 – 2.5	Moderately low	
2.5 - 6.0	Moderate	
>6.0	Very high	

Table 4.1: Classification of soil based on the SOC (Feller and Beare, 1997)

Note. SOC= Soil organic carbon

SOM is usually estimated through a measure of SOC and the different soils can be classified based on SOM matter content (Table 4.2). It is determined by multiplying

the value of organic carbon with 1.72 (Walkley, 1935).

The soil organic matter (SOM) equation is:

SOM = C × 1.72

Whereby: C is the soil organic carbon, expressed in percentage (%).

1.72 is the Bemmelen factor

Table 4.2: Classification of soil based on the SOM content (Feller and Beare, 1997).

SOM level (%)	Description
0.00 - 0.5	Extreme low; soil is deprived of residues
0.5 – 1.0	Low ; soil needs more organic residues
1.0 - 2.5	Moderately low ; soil has been cropped very heavy
2.5 - 6.0	Moderate; soil is being maintained in optimal desired range.
>6.0	Soil is accumulating organic matter in high addition rate; rich organic soil.

Particle size distribution was determined using the hydrometer method (Bouyoucos, 1962). Soil pH was first measured in deionized water (1:2 soil, water) followed by 0.01 M calcium chloride (CaCl₂) using a calibrated glass electrode pH meter (Rhodes, 1982).

4.4.2.2 Remote sensing data

```
Multispectral imagery (sentinel 2)
```

In this study, the Sentinel-2 multispectral imagery was used in assessing SOC. This imagery is able to provide excellent spatial coverage of a large area, and making it easy to obtain soil information (Gorji *et al.*, 2017). The multispectral image is a type of raster data in which raster files are grid of pixels (cells) and each pixel contains a single value (references) that provide valuable information. The multispectral imagery is able to provide valuable information that might not be able to obtain using

conventional methods since our visual perception is limited to some portions of EMS (Electromagnetic spectrum).

a) Image acquisition

A multispectral satellite image of the study area was collected on 09 March 2021. This imagery (Sentinel 2) was downloaded from the European Space Agency (ESA); the image had more than 10 bands ranging from the visible to shortwave infrared and with 10 m spatial resolution.

b) Image processing

Before the multispectral satellite image was used, it went through pre-processing. Then images went through the stages of atmospheric and radiometric correction using Quantum Geographic Information System (QGIS). This was done in order to extract specific or valuable information. The imagery was atmospherically corrected using QGIS.

4.4.3 Data analysis

This study used one regression model (PLSR model) to estimate the relationship or correlation between the corresponding pixels or spectral values extracted from multispectral satellite imagery (Sentinel 2) and conventionally determined SOC. The coordinates of the sampling points were used in QGIS to extract spectral or reflectance values from the image using vector point extraction. PLSR is a method that specifies a linear relationship between a set of dependent variables, Y, and a set of predictor variables, X (Farifteh *et al.*, 2007). It reduces the variables, used to predict, to a smaller set of predictors. The general idea of the PLSR is to extract the orthogonal or latent predictor variables, accounting for as much of the variation of the dependent variables as possible. PLSR was chosen due to its capability to analyse large data, and it is more interpretable (Asfaw *et al.*, 2018). It is an effective tool for assessment of soil degradation forms (Gorji *et al.*, 2017).

Soil organic carbon (SOC) is the dependent variable whereas the spectral or reflectance values from multispectral images are the independent variable. The prediction performance of this model was evaluated based on the coefficient of determination (R^2), root mean square error of prediction (RMSE) and ratio of prediction deviation (RPD). The soil samples were divided into calibration and

validation group. Therefore, 70% was used for training or model development and 30% for testing or model validation.

a) Calibration

Conventional laboratory measurement and their corresponding reflectance were used for calibrating models using partial least square regression (PLSR) to verify the results and test the goodness of the model or performance. Conventional laboratory measurement of SOC for validation was used to test the goodness of the model or performance. PLSR model was developed for estimation of the SOC based on the reflectance or spectral variation.

b) Validation

The Root Mean Square Error (RMSE), coefficient of determination (R^2) and relative percentage deviation (RPD) were used to test the predictive ability or accuracy of the model. RMSE provides the absolute average error between the predicted values and the measured results of conventional laboratory analysis (Leone *et al.*, 2012). The coefficient of determination (R^2) measures the proportion of the total variation accounted for and it has the capability to test the goodness of the model (Table 4.3). RPD refers to the ratio of standard deviation (SD) to the RMSE prediction. It was obtained by dividing the standard deviation of analysed data by the value of RMSE and it was used to test Prediction of ability of the model (Table 4.4). RMSE is used to measure the difference between the values predicted and the values observed. Chang *et al.* (2001) and Mouazen *et al.* (2010) developed a method to classify models based on their estimation accuracy, looking at their RPD value and R^2 value respectively.

Table 4.3: Prediction of goodness of the model based on the coefficient of determination (R^2) supplied by Mouazen *et al.* (2010)

R ²	Model performance
0.5 to 0.65	Poor prediction model
0.5 to 0.79	Good
0.80 to 1.00	Excellent

Note. R^2 = coefficient of determination

Table 4.4: Prediction of goodness of the model based on the relative percentage deviation (RPD) supplied by Duran (1983)



Figure 4.2: Flowchart of methodology used in this study

4.5 RESULTS

4.5.1 Conventional laboratory analysis results for soil physical and chemical properties.

Based on the results obtained there is a great spatial variation of SOC in the study site (Table 4.5). The mean SOC and SOM were found to be 1.66% and 2.86% respectively. The highest SOC was found to be 3.56% while the lowest to be 0.76%. The highest SOM content was found to be 6.12 and the lowest to be 1.30. The low SOM content will results in poor water and nutrient retention. The high SOM content inhibits soil erosion because SOM binds soil particles together. The standard deviation of SOC and SOM is 0.48 and 0.82 respectively. The coefficient of variation was found to be 28.92 and 28.67 percentages for SOC and SOM respectively. The results show that clay content range from 0.18 to 30.31, sand percentage range from 49.86 to 79.40 and silt percentage range from 0.21 to 37.86. This shows that there is variation across the field and may possibly be as a result of human and natural factors.

Soil parameter	Max	Min	Mean	Mediar	n SD	CV (%)
SOC (%)	3.56	0.76	1.66	1.62	0.48	28.92
SOM (%)	6.12	1.30	2.86	2.77	0.82	28.67
Clay (%)	30.31	0.18	18.38	12.30	7.05	38.36
Sand (%)	79.40	49.86	62.32	69.45	6.89	11,06
Silt (%)	37.86	0.21	20.60	16.38	8.34	40.49
pH(KCl)	7.06	6.06	6.31	6.30	0.27	4.28
pH(H ₂ O)	7.49	6.7	6,71	6.72	0.31	4.619

Table 4.5: Statistical description of soil information analysed (n =98) in Syferkuil farm, Limpopo, South Africa

Note. SD = standard deviation; max = maximum; min = minimum; CV = coefficient of variation.

4.5.2 Prediction of soil organic carbon using PLSR model

The results obtained showed the poor estimation accuracy for model performance of SOC when using multispectral satellite image (R^2 of 0.41, RMSE of 0.53 and; RPD of 1.21(Table 4.6). The PLSR model performance for SOC prediction was found to be unsatisfactory. The bands which were used in SOC prediction fall within the visible, shortwave infrared and near-infrared regions of the electromagnetic spectrum.

Table 4.6: PLSR model performance for soil organic carbon prediction

Soil parameter	R ²	RMSE	RPD	Model performance	•
SOC	0 .41	0.53	1.21	Not excellent	

4.6 DISCUSSION

4.6.1 Soil organic carbon and soil degradation forms

The results obtained show that soil organic carbon (SOC) status was lower and higher in some other parts in the study area. The areas of low SOC status (1.5% and less) observed in the study area may have influenced by low elevation, slope and agricultural land use (Phinzi *et al.*, 2020; Mohammed *et al.*, 2020). Therefore, the low SOC status may possibly be a result of frequent deep ploughing which disturb the soils extensively. Taking into considerations that the conventional method (i.e. tillage practice) was practiced in the portion were soil samples collected. Conventional practices breakdown soil aggregates hence leading to depletion of SOM, thus reducing carbon inputs into the soil. However, no tilled soils tend to store more carbon than the one found on similar conventional soils because of the formation of strong structure that limit soil erosion hence increases crop productivity and development. For instance, Bayer *et al.* (2006) have pointed out that those soils under no till method store more carbon due to increased dead matter accumulation and amount of undisturbed biomass. The conventional laboratory analysis results for soil organic carbon and soil organic matter are in Appendix 3.

Conventional methods can also influence SOC status since it will results in soils left exposed and the upper profile be easily removed (Phinzi and Ngetar 2017; and Sepuru and Dube, 2018). This is why vegetation cover is considered as one of the most important factors controlling soil erosion and can considerably decrease soil loss, due to its ability to bind soil particles, thereby protecting the soil (Jain and Goel, 2002). The results of this study can differ with other studies possibly due to the differences in the climatic conditions or rainfall patterns and the period at which tillage system has been in practice. Conventional methods are currently threatened by population growth, agriculture and increased demand for food production. This means that when the ever growing population increases, there is higher demand for food production to feed the human population hence limited time to practice sustainable agriculture. This is supported by Pacione (2013), who pointed out that an increase in the number of human population results in higher demand for food to feed the ever-growing global population and thus accelerate land degradation.

Poor agricultural and land management practices without taking into considerations sustainable conservation practices lead to accelerated soil erosion (Le Roux *et al.*, 2007). This means that the upper portions of the top soil can be easily removed and results in decline in soil organic carbon content. Another possible reason may be due to natural factors (i.e. slope, rainfall and soil type) which lead to the development and initiation of soil degradation forms and results in decline in soil organic carbon status (Suleymanov *et al.*, 2021; Mallik *et al.*, 2022). This is because at the research site the soils are prone to erosion due to permeability or infiltration rate which results in increased surface runoff then remove the upper top layer that is composed of high concentration of organic materials.

The lack of knowledge and resources may also possibly result in decline in SOC status. This is supported by Seutloali *et al.* (2017) who emphasized that poor decisions and mismanagement of soils can be able to accelerate soil degradation forms. This will then lead to inefficient land use planning and management. Population growth is also driving force for changes in land use management. This is supported by Pacione. (2013), who pointed out that an increase in the number of human population results in higher demand for food to feed the ever growing global population and thus accelerate soil degradation forms.

The high SOC status in some portion of study area observed can also be caused by high concentration of clay content as compared to high areas with low clay content and high erosion rates. This is supported by various studies by Ben-Dor, 2002 and Guerschman *et al.* (2009). They found that the high concentration of clay content plays a crucial role in plant productivity and growth. This is because high SOC is associated with high concentration of clay content which then influences water holding capacity, infiltration and nutrients availability. Clay textured soils are more compacted and essential nutrients are not easily leached out during heavy rainfall. Furthermore, high SOC can increase soil fertility and improve yields and food security (Ben-Dor, 2002; Guerschman *et al.*, 2009).

4.6.2 The performance of model developed from multispectral satellite image.

The performance of the model was not excellent. The result shows that approximately half of the SOC variation could be explained by the reflectance values as recorded in the image. The deviation of the predicted values from the PLSR analysis was minimal as indicated by the low RMSE value. The low R² value indicates that there are confounding factors. The confounding factors might come in the form of noise and errors due to variations in soil surface roughness, geometric and atmospheric conditions (Casa et al., 2013). Furthermore, the performance of models might be because of low spectral resolution of the image as compared to hyperspectral data, which have bands with narrow wavelengths (Qi et al., 2020). The results of this study are supported by Allbed and Kumar (2013) who highlighted that multispectral data has limitations because of the coarse spatial and spectral resolutions, which influences the quality and quantity of the information they provide. This may be due to low swath width when using multispectral imagery which does not permits large area coverage and frequent mapping as compared to Land sat imagery (Odindi et al., 2015). However, the performance of the models varies with the study site and the regression techniques used (Gorji et al., 2017; Shahabi et al., 2021).

4.7 CONCLUSION AND RECOMMENDATIONS

This study focused on the use of remote sensing techniques in assessing soil organic carbon status. The extracted reflectance values from Sentinel 2 were used together with results of conventional laboratory analysis results for prediction performance of PLSR models. In this study a multispectral satellite image was used and it did not give satisfactory results. Despite that the performance of the model was unsatisfactory, remote sensing techniques are considered as efficient tools to monitor and detect SOC unlike in the old days, when conventionally laboratory methods were the only means of assessing SOC status even though it produced low prediction accuracy. Thus, it is recommended that an image with better spectral resolution and narrower bands (hyperspectral) be investigated to see if there is an improvement in the model performance. This study can be helpful for land use management and planning and increases awareness on the importance of managing and restoring soil quality to minimise the risk of soil degradation forms.

CHAPTER FIVE

SUMMARY AND RECOMMENDATIONS

The study revealed the potential of using remote sensing as an alternative method to the conventional methods in assessing soil degradation forms. PLSR models developed using spectral reflectance extracted from Sentinel 2 imagery and the conventional laboratory results (i.e., soil organic carbon and soil salinity analysis) did not give satisfactory results in chapter two and four. Based on the results, this might be due to noise and errors due to variations in soil surface roughness, geometric and atmospheric effects. Image noise is any unwanted disturbance in image data that is due to limitations in the sensing, signal digitization, or data recording process. In addition, the performance of PLSR models might be influenced by the spatial and spectral resolution of the image. This is because the multispectral imagery used in this study has low spectral and spatial resolution as compared to higher spatial resolution imagery and narrow bands (hyperspectral satellite image). The satellites imagery with relatively higher spatial and spectral resolution allows the classification technique to detect the smaller rill erosion features which were omitted from low spatial resolution imagery (Qi et al., 2017). In chapter three, the results were not satisfactory from multispectral satellite image [Worldview 2 satellite image] using digital image processing called supervised image classification. Despite low prediction accuracy, the potential of remote sensing techniques are evident. Remote sensing has the potential to serve as an alternative tool in monitoring and identifying soils degraded by rill erosion and could reduce the field based work.

Based on the results obtained and with the current demand for valuable or precise up to date soil degradation forms, remote sensing is a crucial or effective tool. It can be utilized with the upcoming studies and it has the potential to identify and monitor soil degradation form. Thus, it is recommended that an image with a better spatial and spectral resolution (i.e. hyperspectral remote sensing data) be investigated to see if there is improvement in the model performance. This is because in remote sensing application there is much room for improvement, and can be valuable particularly in data scarce environment like Limpopo. Lastly, it is also recommended that the entire area should be put under careful monitoring, and emphasis should be given to sustainable land management measures.

REFERENCES

- Acosta, J.A., Jansen, B., Kalbitz, K., Faz, A. and Martínez-Martínez, S. 2011. Salinity increases mobility of heavy metals in soils. *Chemosphere 85(8)*: 1318-1324.
- Aggarwal, S. 2004. Principles of remote sensing. *Satellite Remote Sensing and GIS Applications in Agricultural Meteorology*. 23-38.
- Aksoy, S., Yildirim, A., Gorji, T., Hamzehpour, N., Tanik, A. and Sertel, E. 2022. Assessing the performance of machine learning algorithms for soil salinity mapping in Google Earth Engine platform using Sentinel-2A and Landsat-8 OLI data. *Advances in Space Research 69(2*): 1072-1086.
- Aldabaa, A.A.A., Weindorf, D.C., Chakraborty, S., Sharma, A. and Li, B. 2015. Combination of proximal and remote sensing methods for rapid soil salinity quantification. *Geoderma 239*: 34-46.
- Al-Gaadi, K.A., Tola, E., Madugundu, R. and Fulleros, R.B. 2021. Sentinel-2 images for effective mapping of soil salinity in agricultural fields. *Current Science 121*: 384-390.
- Allbed, A. and Kumar, L. 2013. Soil salinity mapping and monitoring in arid and semiarid regions using remote sensing technology: A Review. *Advances in Remote Sensing 2013 (2): 373 - 385*
- Angelopoulou, T., N. Tziolas, A. Balafoutis, G. Zalidis, and D. Bochtis. 2019. Remote Sensing Techniques for Soil Organic Carbon Estimation: A Review. *Remote Sensing 11 (6):* 676.
- Asfaw, E., Suryabhagavan, K.V. and Argaw, M. 2018. Soil salinity modelling and mapping using remote sensing and GIS: The case of Wonji sugar cane irrigation farm, Ethiopia. *Journal of the Saudi Society of Agricultural Sciences 17*: 250-258.
- Bai, Z. and Dent, D. 2009. Recent land degradation and improvement in China. *Ambio*: 150-156.
- Bangelesa, F., Adam, E., Knight, J., Dhau, I., Ramudzuli, M. and Mokotjomela, T.M. 2020. Predicting soil organic carbon content using hyperspectral remote sensing in a degraded mountain landscape in Lesotho. *Applied and Environmental Soil Science 2020 (1):* 1 -11.
- Bangroo, S.A., Najar, G.R., Achin, E. and Truong, P.N. 2020. Application of predictor variables in spatial quantification of soil organic carbon and total nitrogen

using regression kriging in the North Kashmir forest Himalayas. *Catena 193*: 104632.

- Ben-Dor, E. Patkin, K. Banin, A., and Karnieli, A. 2002. Mapping of several soil properties using DAIS-7915 hyperspectral scanner data. *International Journal of Remote Sensing* 23:1043 -1062.
- Casa, R., Castaldi, F., Pascucci, S., Palombo, A. and Pignatti, S. 2013. A comparison of sensor resolution and calibration strategies for soil texture estimation from hyperspectral remote sensing. *Geoderma 197*: 17-26.
- Castaldi, F., Chabrillat, S., Don, A. and van Wesemael, B. 2019. Soil organic carbon mapping using LUCAS topsoil database and Sentinel-2 data: An approach to reduce soil moisture and crop residue effects. *Remote Sensing 11*(18): 2121.
- Chaplot, V. 2013. Impact of terrain attributes parent material and soil types on gully erosion. *Geomorphology 186*: 1-11.
- Chauhan, B. 2015. Agricultural and hydrological applications of remote sensing. India: Random publications
- Chen, K., Chen, H., Zhou, C., Huang, Y., Qi, X., Shen, R., Liu, F., Zuo, M., Zou, X., Wang, J. and Zhang, Y. 2020. Comparative analysis of surface water quality prediction performance and identification of key water parameters using different machine learning models based on big data. *Water Research 171*: 115454.
- Conforti, M., Froio, R., Matteucci, G. and Buttafuoco, G. 2015. Visible and near infrared spectroscopy for predicting texture in forest soil: An application in Southern Italy. *Biogeosciences and Forestry 8*: 339-347
- Congedo, L. 2016. Semi-automatic classification plugin documentation. *Release 4*(1): 29.
- Curcio, D., Ciraolo, G., D'Asaro, F. and Minacapilli, M. 2013. Prediction of soil texture distributions using VNIR-SWIR reflectance spectroscopy. *Procedia Environmental Sciences* 19: 494-503.
- Data, P.D.H., Jakob, S., Zimmermann, R. and Gloaguen, R. 2017. The Need for Accurate Geometric and Radiometric Corrections of Drone-Borne Hyperspectral Data for Mineral Exploration. *Remote Sensing 9:*
- de Santana, F.B., de Souza, A.M. and Poppi, R.J. 2018. Visible and near infrared spectroscopy coupled to random forest to quantify some soil quality parameters. *Molecular and Biomolecular Spectroscopy* 191: 454-462.

DeLong, C., Cruse, R. and Wiener, J. 2015. The soil degradation paradox:

Compromising our resources when we need them the most. *Sustainability* 7: 866-879.

- Desprats, J.F., Raclot, D., Rousseau, M., Cerdan, O., Garcin, M., Le Bissonnais, Y., Ben Slimane, A., Fouché, J. and Monfort-Climent, D. 2013. Mapping linear erosion features using high and very high resolution satellite imagery. *Land Degradation and Development 24*(1): 22-32.
- Dewitte, O., Daoudi, M., Bosco, C. and Van Den Eeckhaut, M. 2015. Predicting the susceptibility to gully initiation in data-poor regions. *Geomorphology 228*: 101-115.
- Dwivedi, R.S. 2001. Soil resources mapping: A remote sensing perspective. *Remote Sensing Reviews 20*(2): 89-122.
- Eastman, J.R. 2003. IDRISI Kilimanjaro: Guide to GIS and image processing. Clark University. USA.
- Farifteh, J., Van der Meer, F., Atzberger, C. and Carranza, E.J.M. 2007. Quantitative analysis of salt-affected soil reflectance spectra: A comparison of two adaptive methods (PLSR and ANN). *Remote Sensing of Environment 110(1):* 59-78.
- Fiorucci, F., Ardizzone, F., Rossi, M. and Torri, D. 2015. The use of stereoscopic satellite images to map rills and ephemeral gullies. *Remote Sensing 7*(10): 14151-14178.
- Floras, S.A. and Sgouras, I.D. 1999. Use of geo-information techniques in identifying and mapping areas of erosion in a hilly landscape of central Greece. *International Journal of Applied Earth Observation and Geoinformation* : 68–77.
- Franceschini, M.H.D., Dematte, J.A.M., da Silva Terra, F., Vicente, L.E., Bartholomeus,
 H. and de Souza Filho, C.R. 2015. Prediction of soil properties using imaging spectroscopy: Considering fractional vegetation cover to improve accuracy. *International Journal of Applied Earth Observation and Geoformation* 38: 358-370.
- Franzluebbers, A.J. and Stuedemann, J.A. 2002. Particulate and non-particulate fractions of soil organic carbon under pastures in the Southern Piedmont USA. *Environmental Pollution 116*: 53- 62.
- Gafurov, A. 2022. Mapping of Rill Erosion of the Middle Volga (Russia) Region Using Deep Neural Network (DNN). *ISPRS International Journal of Geo*-
Information 11(3): 197.

- Gao, Y., Shao, G., Wu, S., Xiaojun, W., Lu, J. and Cui, J. 2021. Changes in soil salinity under treated wastewater irrigation: A meta-analysis. *Agricultural Water Management 255*: 106986.
- Gastellu-Etchegorry, J.P., Lauret, N., Yin, T., Landier, L., Kallel, A., Malenovský, Z., Al Bitar, A., Aval, J., Benhmida, S., Qi, J. and Medjdoub, G. 2017. DART: recent advances in remote sensing data modeling with atmosphere, polarization, and chlorophyll fluorescence. *IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing* 10(6): 2640-2649.
- Gholizadeh, H., Gamon, J.A., Zygielbaum, A.I., Wang, R., Schweiger, A.K. and Cavender-Bares, J. 2018. Remote sensing of biodiversity: Soil correction and data dimension reduction methods improve assessment of diversity (species richness) in prairie ecosystems. *Remote Sensing of Environment 206*: 240-253.
- Gopalakrishnan, T. and Kumar, L. 2020. Modeling and mapping of soil salinity and its impact on Paddy Lands in Jaffna Peninsula, Sri Lanka. *Sustainability 12*: 8317.
- Gorji, T., Sertel, E. and Tanik, A. 2017. Monitoring soil salinity via remote sensing technology under data scarce conditions: A case study from Turkey. *Ecological Indicators 74*: 384-391.
- Gorji, T., Yildirim, A., Sertel, E. and Tanik, A. 2019. Remote sensing approaches and mapping methods for monitoring soil salinity under different climate regimes. *International Journal of Environment and Geo informatics 6*: 33-49.
- Guerschman, J.P., Hill, M.J., Renzullo, L.J., Barrett, D.J., Marks, A.S. and Botha, E.J. 2009. Estimating fractional cover of photosynthetic vegetation, nonphotosynthetic vegetation and bare soil in the Australian tropical savanna region upscaling the EO-1 Hyperion and MODIS sensors. *Remote Sensing of Environment 113*(5): 928-945.
- Hailu, B. and Mehari, H. 2021. Impacts of Soil Salinity/Sodicity on Soil-Water Relations and Plant Growth in Dry Land Areas: A Review. *Journal of Natural Sciences 12*(3): 1-10.
- Houghton, R.A. 2007. Balancing the global carbon budget. *Annual Review of Earth and Planetary Sciences 35*(1): 313-347.
- Iwai, C.B., Oo, A.N. and Topark-ngarm, B. 2012. Soil property and microbial activity in natural salt affected soils in an alternating wet dry tropical climate.

Geoderma189: 144-152.

- Jain, S.K. and Goel, M.K. 2002. Assessing the vulnerability to soil erosion of the Ukai Dam catchments using remote sensing and GIS. *Hydrological Sciences Journal 47*(1): 31- 40.
- Kakembo, V., Xanga, W.W. and Rowntree, K. 2009. Topographic thresholds in gully development on the hillslopes of communal areas in Ngqushwa Local Municipality, Eastern Cape, South Africa. *Geomorphology 110*:188–194.
- Karydas, C., Bouarour, O. and Zdruli, P. 2020. Mapping spatio-temporal soil erosion patterns in the Candelaro River Basin, Italy, using the G2 model with Sentinel2 imagery. *Geosciences 10*(3): 89.
- Kumar, P. and Sharma, P.K. 2020. Soil salinity and food security in India. *Frontiers in Sustainable Food Systems 4*: 533781.
- Kumar, P., P. C. Pandey, B. K. Singh, S. Katiyar, V. P. Mandal, M. Rani, V. Tomar, and S. Patairiya. 2016. Estimation of Accumulated Soil Organic Carbon Stock in Tropical Forest Using Geospatial Strategy. *The Egyptian Journal of Remote Sensing and Space Science 19 (1):* 109-123.
- Lal, R. 2015. Restoring soil quality to mitigate soil degradations. *Sustainability 7*: 5875-5895.
- Le Roux, J. J. and Sumner, P. 2012. Factors controlling gully development: comparing continuous and discontinuous gullies. *Land Degradation and Development 23*: 440- 449.
- Le, Q.B., Nkonya, E. and Mirzabaev, A. 2016. Biomass productivity-based mapping of global land degradation hotspots. *Economics of land degradation and improvement–A global assessment for sustainable development 55*.
- Leone, A.A, Viscarra-Rossel, R., Amenta, P. and Buondonno, A. 2012. Prediction of soil properties with PLSR and vis-NIR spectroscopy: Application to mediterranean soils from Southern Italy. *Current Analytical Chemistry 8*(2): 283-299.
- Levi, N., Hillel, N., Zaady, E., Rotem, G., Ziv, Y., Karnieli, A. and Paz- Kagan, T. 2021. Soil quality index for assessment of phosphate mining restoration in a hyperarid environment. *Ecological indicators* 125: 107571.
- Li, Z. and Fang, H. 2016. Impacts of climate change on water erosion: A review. Earth -Sci. *Rev 163*: 94–117.

Liao, K., Xu, S., Wu, J. and Zhu, Q. 2013. Spatial estimation of surface soil texture

using remote sensing data. *Soil Science and Plant Nutrition* 59(4): 488-500.

- Lillesand, T.M., Kiefer, R.W. and Chipman, J.W. 2008. Remote Sensing and image interpretation, 6th edition, John Wiley and Sons, Hoboken.
- Mallik, S., Bhowmik, T., Mishra, U. and Paul, N., 2022. Mapping and prediction of soil organic carbon by an advanced geostatistical technique using remote sensing and terrain data. *Geocarto International*, *37*(8), pp.2198-2214.
- Mararakanye, N. 2015. A comparative study of gully erosion contributing factors in two tertiary catchments in Mpumalanga South Africa (Doctoral dissertation, University of Pretoria). p 15 48.
- Mararakanye, N. and Le Roux, J.J. 2011. Manual digitising of gully erosion in South Africa using high resolution SPOT 5 satellite imagery at 1: 10 000 scale.
- Mararakanye, N. and Sumner, P.D. 2017. Gully erosion: A comparison of contributing factors in two catchments in South Africa. *Geomorphology 28:* 99–110.
- Mather, P.M. 2004. Computer processing of Remotely-Sensed images: an introduction, 3rd edition, John Wiley and Sons, Chichester
- McCauley, A., Jones, C. and Olson-Rutz, K. 2017. Soil pH and Organic Matter. Nutrient Management Module No. 8. Montana State University. USA.
- Metternicht, G.I. and Zinck, J.A. 2003. Remote sensing of soil salinity: potentials and constraints. *Remote Sensing of Environment 85*: 1-20.
- Minu, S., Shetty, A., Minasny, B. and Gomez, C., 2017. The role of atmospheric correction algorithms in the prediction of soil organic carbon from Hyperion data. *International Journal of Remote Sensing*, *38*(23), pp.6435-6456.
- Molepo, K.J., Ekosse, G.I.E. and Ngole-Jeme, V.M. 2017. Physicochemical, geochemical and mineralogical aspects of agricultural soils in Limpopo Province, South Africa. *Journal of Human Ecology 58*(1-2): 108-117.
- Morshed, M., Islam, M. and Jamil, R. 2016. Soil salinity detection from satellite image analysis: an integrated approach of salinity indices and field data. *Environmental Monitoring and Assessment 188*(2): 1-10.
- Mouazen, A.M., Kuang, B., De Baerdemaeker, J. and Ramon, H. 2010. Comparison among principal component, partial least squares and back propagation neural network analyses for accuracy of measurement of selected soil properties with visible and near infrared spectroscopy. *Geoderma 158(1-2)*: 23 -31.

Mucina, L. and Rutherford, M.C. 2006. The vegetation of South Africa, Lesotho and

Swaziland. Sterlitzia 19. South African National Biodiversity Institute, Pretoria.

- Nabiollahi, K., Eskandari, S., Taghizadeh-Mehrjardi, R., Kerry, R. and Triantafilis, J. 2019. Assessing soil organic carbon stocks under land-use change scenarios using random forest models. *Carbon Management 10*(1): 63-77.
- Nkonya, E., Mirzabaev, A. and Von Braun, J. 2016. Economics of land degradation and improvement, a global assessment for sustainable development. p 686.
- Nocita, M., Stevens, A., Toth, G., Panagos, P., van Wesemael, B. and Montanarella, L. 2014. Prediction of soil organic carbon content by diffuse reflectance spectroscopy using a local partial least square regression approach. *Soil Biology and Biochemistry 68*: 337-347.
- Nwilo, P.C., Ogbeta, C.O., Daramola, O.E., Okolie, C.J. and Orji, M.J. 2021. Soil Erosion Susceptibility Mapping of Imo River Basin Using Modified Geomorphometric Prioritisation Method. *Quaestiones Geographicae 40*(3): 143-162.
- Odindi, J., Mutanga, O., Abdel-Rahman, E. M., Adam, E. and Bangamwabo, V. 2017. Determination of urban land-cover types and their implication on thermal characteristics in three South African coastal metropolitans using remotely sensed data. *South African Geographical Journal 99(1)*: 52-67.
- Odindi, J., O. Mutanga, M. Rouget, and N. Hlanguza. 2016. Mapping Alien and Indigenous Vegetation in the KwaZulu-Natal Sandstone Sourveld Using Remotely Sensed Data. *Bothalia 46 (2)*: 1-9.
- Ogbonna, J.U. 2012. Understanding gully erosion vulnerability in Old Imo State using geographic information system and geostatistics. *American Journal of Geographic Information System 1*(3): 66-71.
- Pacione, M. 2013. *Historical geography: progress and prospect*. Routledge.
- Paterson, G., Van Ziyl, G., Van Tol, J., Turner, D., Wiese, L. and Clarke, C. 2015. Spatial soil information in South Africa: Situational analysis: limitations and challenges. *South African Journal of Science* 111: 1-7.
- Peng, J., Biswas, A., Jiang, Q., Zhao, R., Hu, J., Hu, B. and Shi, Z. 2019. Estimating soil salinity from remote sensing and terrain data in southern Xinjiang Province, China. *Geoderma 337*: 1309-1319.
- Pennock, D., McKenzie, N. and Montanarella, L. 2015. Status of the world's soil resources. Technical Summary FAO, Rome, Italy.
- Persello, C. and Bruzzone, L. 2010. A novel protocol for accuracy assessment in classification of very high resolution images. *IEEE Transactions on*

Geoscience and Remote Sensing 48(3): 1232–1244.

- Phefadu, K.C. and Kutu, F.R. 2016. Evaluation of spatial variability of soil physicochemical characteristics on Rhodic Ferralsol at the Syferkuil Experimental Farm of University of Limpopo, South Africa. *Journal of Agricultural Science 8*(10): 92.
- Phinzi, K. and Ngetar, N.S. 2017. Mapping soil erosion in a quaternary catchment in Eastern Cape using geographic information system and remote sensing. *South African Journal of Geomatics 6*(1): 11-29.
- Qi, X., Zhu, P., Wang, Y., Zhang, L., Peng, J., Wu, M., Chen, J., Zhao, X., Zang, N. and Mathiopoulos, P.T. 2020. A multi-label high spatial resolution remote sensing dataset for semantic scene understanding. *ISPRS Journal of Photogrammetry and Remote Sensing 169*: 337-350.
- Rahmati, O. and Jaafari, A. 2021. Spatial Modeling of Soil Erosion Susceptibility with Support Vector Machine. In *Intelligent Data Analytics for Decision-Support Systems in Hazard Mitigation*: 267-280.
- Ren Wang, Z., Wang, G., Ren, T., Wang, H., Xu, Q. and Zhang, G. 2021. Assessment of soil fertility degradation affected by mining disturbance and land use in a coalfield via machine learning. *Ecological Indicators* 125: 107608.
- Ren, D., Wei, B., Xu, X., Engel, B., Li, G., Huang, Q., Xiong, Y. and Huang, G. 2019. Analysing spatiotemporal characteristics of soil salinity in arid irrigated agroecosystems using integrated approaches. *Geoderma 356*:113935.
- Rossel, R.A, Viscarra. and Chen, C. 2011. Digitally mapping the information content of visible-near infrared spectra of surficial Australian soils. *Remote Sensing of Environment 115(6)*: 1443-1455.
- Saadat, H., Adamowski, J., Tayefi, V., Namdar, M., Sharifi, F. and Ale-Ebrahim, S. 2014. A new approach for regional scale interrill and rill erosion intensity mapping using brightness index assessments from medium resolution satellite images. *Catena 113*: 306-313.
- Samani, A.N., Ahmadi, H., Jafari, M., Boggs, G., Ghoddousi, J. and Malekian, A. 2009. Geomorphic threshold conditions for gully erosion in Southwestern Iran. *Journal of Asian Earth Sciences 35*:180-189.
- Sepuru, T. K. and Dube, T. 2018. Understanding the spatial distribution of eroded areas in the former rural homelands of South Africa: Comparative evidence from two new non- 59 commercial multispectral sensors. *International*

Journal of Applied Earth Observation and Geoinformation 69: 119-132.

- Seutloali, K. E., Beckedahl, H. R, Dube, T. and Sibanda, M. 2016. An assessment of gully erosion along major armoured roads in south-eastern region of South Africa: a remote sensing and GIS approach. *Geocarto International 31:* 225-239.
- Seutloali, K.E., Dube, T. and Mutanga, O. 2017. Assessing and mapping the severity of soil erosion using the 30-m Landsat multispectral satellite data in the former South African homelands of Transkei. *Physics and Chemistry of the Earth, Parts A/B/C, 100*, pp.296-304.
- Shahabi, H., Rahimzad, M., Tavakkoli Piralilou, S., Ghorbanzadeh, O., Homayouni, S., Blaschke, T., Lim, S. and Ghamisi, P. 2021. Unsupervised deep learning for landslide detection from multispectral sentinel-2 imagery. *Remote Sensing*, *13*(22): 4698.
- Shahabi, M., Jafarzadeh, A.A., Neyshabouri, M.R., Ghorbani, M.A. and Valizadeh Kamran, K. 2017. Spatial modeling of soil salinity using multiple linear regression, ordinary kriging and artificial neural network methods. *Archives of Agronomy and Soil Science 63*(2): 151-160.
- Shan, W., Jin, X., Ren, J., Wang, Y., Xu, Z., Fan, Y., Gu, Z., Hong, C., Lin, J. and Zhou, Y.
 2019. Ecological environment quality assessment based on remote sensing data for land consolidation. *Journal of Cleaner Production 239*: 118126.
- Shen, H., Li, X., Cheng, Q., Zeng, C., Yang, G., Li, H. and Zhang, L. 2015. Missing information reconstruction of remote sensing data: A technical review. *IEEE Geoscience and Remote Sensing Magazine 3*(3): 61-85
- Shi, P., Zhang, Y., Li, P., Li, Z., Yu, K., Ren, Z., Xu, G., Cheng, S., Wang, F. and Ma, Y. 2019. Distribution of soil organic carbon impacted by land-use changes in a hilly watershed of the Loess Plateau, China. *Science of the Total Environment* 652: 505-512.
- Shrestha, R.P. 2006. Relating soil electrical conductivity to remote sensing and other soil properties for assessing soil salinity in northeast Thailand. *Land Degradation and Development 17*(6): 677- 689.
- Shruthi, R.B., Kerle, N. and Jetten, V. 2011. Object-based gully feature extraction using high spatial resolution imagery. *Geomorphology 134*(3-4): 260-268.
- Sonneveld, M.P.W., Everson, T.M. and Veldkamp, A. 2005. Multi-scale analysis of soil erosion dynamics in Kwazulu-Natal, South Africa. *Land Degradation and*

Development 16: 287-301.

- Steinberg, A., Chabrillat, S., Stevens, A., Segl, K. and Foerster, S. 2016. Prediction of common surface soil properties based on Vis-NIR airborne and stimulated EnMap imaging spectroscopy Data: Predictions accuracy and influence of spatial resolutions. *Remote Sensing Journal 8*: 618.
- Stenberg, B., Viscarra Rossel, R.A., Mouazen, A.M. and Wetterlind, J. 2010. Visible and near infrared spectroscopy in soil science: *Advances in Agronomy 107*: 163 - 215.
- Suleymanov, A., Gabbasova, I., Suleymanov, R., Abakumov, E., Polyakov, V. and Liebelt, P. 2021. Mapping soil organic carbon under erosion processes using remote sensing. *Hungarian Geographical Bulletin 70*(1): 49-64.
- Taghadosi, M.M., Hasanlou, M. and Eftekhari, K. 2019. Retrieval of soil salinity from Sentinel-2 multispectral imagery. *European Journal of Remote Sensing 52*(1): 138-154.
- Taruringa, K. 2009. Gully mapping using remote sensing: Case study in KwaZulu-Natal, South Africa. Master's thesis, University of Waterloo.
- Torkashvand, A.M. and Alipour, H.R. 2009. Investigation of the possibility to prepare supervised classification map of gully erosion by RS and GIS. *International Journal of Geological and Environmental Engineering 3(7)*: 203-205.
- UNEP, U.D., FAO .2012. SIDS-Focused Green Economy: An Analysis of Challenges and Opportunities. Nairobi: United Nations Environment Programme.
- Vanmaercke, M., Panagos, P., Vanwalleghem, T., Hayas, A., Foerster, S., Borrelli, P., Rossi, M., Torri, D., Casali, J. and Borselli, L. 2021. Measuring, modelling and managing gully erosion at large scales: A state of the art. A Review. *Environmental Earth Sciences*: 218.
- Vaudour, E., Gomez, C., Loiseau, T., Baghdadi, N., Loubet, B., Arrouays, D., Ali, L. and Lagacherie, P. 2019. The impact of acquisition date on the prediction performance of topsoil organic carbon from Sentinel-2 for croplands. *Remote Sensing 11*(18): 2143.
- Viscarra Rossel, R.S., Cattle, S.R., Ortega, A. and Fouad, Y. 2009. In situ measurements of soil colour, mineral composition and clay content by vis-NIR spectroscopy. *Geoderma 150 (3-4)*: 253-266.
- Walkley, A. 1935. An examination of methods for determining organic carbon and nitrogen in soils. *Journal of Agricultural Science 25*: 598-609.

- Wang, J., He, T., Lv, C., Chen, Y. and Jian. W. 2010. Mapping soil organic matter based on land degradation spectral response units using Hyperion images. *International Journal of Applied Earth Observation and Geoinformation* 12: 171-180.
- Wang, W., Chen, S. and Qu, G. 2008. Incident detection algorithm based on partial least squares regression. *Transportation Research Part C. Emerging Technologies 16*: 54-70.
- Wenjun, J., Zhou, S., Jingyi, H. and Shuo, L. 2014. In situ measurement of some soil properties in paddy soil using visible and near-infrared spectroscopy. *PloS one 9*(8): 105708.
- Xie, Y., Lin, H., Ye, Y. and Ren, X. 2019. Changes in soil erosion in cropland in China, 176, 410 418.
- Xu, D., Ma, W., Chen, S., Jiang, Q., He, K. and Shi, Z. 2018. Assessment of important soil properties related to Chinese soil taxonomy based on Vis-NIR reflectance spectroscopy. Computers and Electronics in Agriculture 144:1-4.
- Xu, G., Lu, K., Li, Z., Li, P., Wang, T. and Yang, Y. 2015. Impact of soil and water conservation on soil organic carbon content in a catchment of the middle Han River, China. *Environmental Earth Sciences 74*(8): 6503-6510.
- Zarei, A., Hasanlou, M. and Mahdianpari, M. 2021. A comparison of machine learning models for soil salinity estimation using multi-spectral earth observation data. *ISPRS Annals of Photogrammetry, Remote Sensing and Spatial Information Sciences 3.*
- Zhang, W., Xue, X., Peng, F., You, Q., and Hao, A. 2019. Meta-analysis of the effects of grassland degradation on plant and soil properties in the alpine meadows of the Qinghai, Tibetan Plateau. *Global Ecology and Conservation* 20: 00774.
- Zhou, L., Wei, H., Li, H., Zhao, W., Zhang, Y. and Zhang, Y. 2020. Arbitrary-oriented object detection in remote sensing images based on polar coordinates. *IEEE Access 8*: 223373-223384.
- Zhou, W., Shao, Z., Diao, C. and Cheng, Q. 2015. High resolution remote sensing imagery retrieval using sparse features by auto encoder. *Remote sensing letters 6*: 775-783.
- Zhu, J., Wu, W. and Liu, H.B. 2018. Environmental variables controlling soil organic carbon in top- and subsoils in karst region of southwestern China. *Ecological Indicators* 90: 624- 632.

APPENDICES

	5	•	•
Error Matrix Code	Reference	Classified	Pixel Sum
4	1	1	1
5	1	2	1
4	2	1	9
7	2	2	9

APPENDIX 1: Showing Error Matrix Code, Reference, Classified and Pixel Sum values

APPENDIX 2: Showing Error matrix (pixel count values)

> ERROR MAT	RIX (pixel	count)		
> Reference				
V_Classified	0	1	2	Total
0	0	0	0	0
1	0	1	9	10
2	0	1	9	10
Total	0	2	18	20

APPENDIX 3: Conventional laboratory analysis results for soil organic carbon and soil organic matter.

Soil	Organic	Organic
samples	C%	matter %
A1	1.87	3.22
A2	1.57	2,70
A3	2,29	3.94
A4	1.25	2.14
A5	1.56	2.68
B1	1.26	2.16
B2	2.25	3.88
B3	1.95	3.36
B4	1.64	2.82
B5	1.23	2.12
C1	2.47	4.25
C2	1.63	2.80
C3	1.27	2.19
C4	2.23	3.83
C5	1.65	2.85
D1	1.51	2.60
D2	1.61	2.76
D3	1.50	2.58
D4	2.17	3.73

Soil	Organic	Organic	
samples	C%	matter %	
D5	1.59	2.74	
F1	1.63	2.80	
F2	2.00	3.44	
F3	1.52	2.62	
F4	1.02	1.88	
F5	1.63	2.80	
C1	1.05	2.00	
	1.05	0.10 0.01	
02	1.04	2.31	
63		2.70	
64		2.60	
65	1.44	2.47	
HI	1.69	2.91	
H2	1.8/	3.22	
H3	1.95	3.36	
H4	1.62	2.78	
H5	1.82	3.13	
11	2.10	3.61	
12	1.55	2.66	
13	1.40	2.41	
14	231	398	
15	2.59	445	
10	2.05	1.40	
12	1 75	2 01	
JZ 12	1.75	3.01 2.50	
J3 1/1		2,30	
KI I-O	1.14	1.90	
K2	1.99	3.42	
K3	2.31	3.98	
_L1	2.39	4.10	
L2	1.58	2.72	
L3	1.51	2.60	
M1	1.63	2.80	
M2	1.53	2.64	
M3	2.25	3.88	
N1	1.16	2.00	
N2	1.51	2.60	
N3	175	3 01	
01	0.91	1.57	
02	1 32	2.27	
02	1.52	2.27	
D1	0.08	1.60	
D2	1.60	2.01	
	1 71	2.91	
P3	1./1	2.95	
_P4		1./5	
01	1.5/	2.70	
02	1.51	2.60	
03	1.93	3.32	
Q4	1.79	3.07	
R1	1.03	1.77	
R2	1.39	2.39	
R3	1.38	2.37	
R4	2.86	4.93	
S1	1.51	2.60	
S2	165	285	
<u>S</u> 3	1 99	3.42	
<u>S4</u>	1 30	2 30	
T1	1.05	2.09	
T2	2.56	612	
T2	2.00	277	
13	2.19	J.//	

Soil	Organic	Organic
samples	C%	matter %
Τ4	1.88	3.24
U1	1,27	2,19
U2	1.17	2.02
U3	0.99	1.71
U4	1.15	1.98
J-1	0.92	1.59
J-2	1.15	1.98
K-1	2.23	3.83
K-2	1.73	2.97
L-1	0.90	1.55
L-2	1.44	2.47
M-1	2.35	4.04
M-2	2.54	4.37
N-1	2.06	3.55
N-2	1.94	3.34
0-1	0.76	1.30
0-2	1.59	2,74
F-1	0.96	1.65
G-1	1.88	3.24
H-1	0.91	1.57
1-1	146	2 52